11/28/06

Budd-Falen Law Offices, L.L.

Karen Budd-Falen Franklin J. Falen¹ Marc R. Stimpert^{1, 2} Brandon L. Jensen^{1, 3} Hertha L. Lund4 Erin Sass Eastman¹ Kathryn Brack Morrow^{3,5}

300 East 18th Street Post Office Box 346 Cheyenne, Wyoming 82003-0346 Telephone 307/632-5105 Telefax 307/637-3891 main@buddfalen.com www.buddfalen.com

ladmitted in Wyoming ²admitted in Oklahoma ³admitted in Colorado ⁴admitted in Montana ⁵admitted in New Mexico

November 28, 2006

Via Federal Express Helen M. Hankins Elko Bureau of Land Management 3900 East Idaho Street Elko, NV 89801

> Re: NOTICE OF APPEAL AND STATEMENT OF REASONS OF THE FINAL GRAZING MANAGEMENT DECISION AND RECORD OF DECISION FOR THE OWHYEE GRAZING ALLOTMENT

Dear Ms. Hankins:

I. NOTICE OF APPEAL

Pursuant to 43 C.F.R. § 4160.4, on behalf of Ellison Ranching Co. ("Ellison," "Appellant" or "permittee"), the purpose of this letter is to appeal the Final Grazing Management Decision and Record of Decision ("ROD") for the Sheep Complex, Big Springs and Owyhee Grazing allotments dated October 30, 2006, file code 1793/4130 (NV-012). The ROD substantially implements the preferred action in the Final Environmental Impact Statement for the Sheep Complex, Big Springs and Owhyee Grazing Allotments Sensitive Bird Species, FES 06-14, dated May, 2006 ("FEIS"). Both the ROD and FEIS were signed by Helen M. Hankins, Manager, Elko Field Office, Bureau of Land Management ("BLM"). This Notice of Appeal and Statement of Reasons pertains ONLY to the portion of the ROD relating to the Owhyee grazing allotment.

II. STATEMENT OF REASONS

A. Jurisdiction of the Office of Hearings and Appeals

The jurisdiction of the Office of Hearings and Appeals is limited to appeals filed by "any person whose interest is adversely affected by a final decision of the authorized officer." See 43

C.F.R. § 4.470(a); see also 43 C.F.R. § 4160.4 (reciting the exact same language); and National Wildlife Federation v. BLM, 145 IBLA 379, 381–82 (1998). Departmental regulation 43 C.F.R. § 4.470(a) confers that "[a]ny applicant, permittee, lessee, or other person whose interest is adversely affected by a final BLM grazing decision may appeal the decision to an administrative law judge." See 43 C.F.R. § 4.470(a). The requirement that there be a "decision of an officer" announcing or prohibiting a specific action before there can be an appeal is essential. See National Wildlife Federation v. BLM, 145 IBLA 379, 382 (1998). The "decision" referred to by the regulation has been interpreted to mean that some action affecting individuals having interests in the public lands is either announced or prohibited. See id. (citing Joe Trow, 119 IBLA 388, 392 (1991)).

In this case, the BLM's October 30, 2006 decision affects the Appellant's right to graze livestock on the public lands and takes a specific action. Ellison is the grazing permittee for the Owhyee allotment. As fully described below, the decision implements a strict rotation system governing the permittee's use of the allotment, prohibits the use of certain pastures in certain situations and imposes utilization standards which will force the permittee to remove his livestock from the allotment in certain situations. For these reasons, it constitutes an appealable decision. See National Wildlife Federation v. BLM, 145 IBLA 379, 382 (1998).

B. Standard of Review

Review of this appeal is governed by the Administrative Procedures Act ("APA"). Under the APA, "[w]hen an application is made for a license required by law, the agency, with due regard for the rights and privileges of all interested parties or adversely affected persons and within a reasonable time, shall set and complete proceedings required to be conducted in accordance with sections 556 and 557 of this title or other proceedings required by law and shall make its decision." 5 U.S.C. § 558(c). A grazing permit issued by the BLM is a license within the meaning of § 558(c). Anchustegui v. Dep't of Agric., 257 F.3d 1124, 1129 (9th Cir. 2001). Therefore, the agency proceedings in this case are governed by § 556 of the APA.

The APA provides that "the proponent of a rule or order has the burden of proof." See 5 U.S.C. §556(d). The United States Supreme Court held that section 7(c) of the APA requires the proponent of the rule or order to meet its burden by a preponderance of the evidence. See Steadman v. S.E.C., 450 U.S. 91, 102 (1981); see also 5 U.S.C. § 556(d) (a rule or order may not be imposed except on consideration of the whole record and supported by and in accordance with the reliable, probative, and substantial evidence). In this case, the BLM is the "proponent of the rule or order" and must prove its decision by a preponderance of the evidence. See John L. Falen, 143 IBLA 1, 4 (1998); and David and Bonnie Ericsson, 88 IBLA 248, 255 (1998).

A BLM decision concerning grazing preference may be considered to be arbitrary, capricious, or inequitable where it is not supported by any rational basis or if it does not substantially comply with the grazing regulations. See Filippini Ranching Co. and Paris Ranch v. BLM, 149 Interior Decision 54, 78 (1999); Riddle Ranches, Inc. v. BLM, 138 IBLA 82, 84 (1997); Kelly v. BLM, 131 IBLA 146, 151 (1994); and Yardley v. BLM, 123 IBLA 80, 90 (1992); Joe Saval Co. v. BLM, 119

IBLA 202, 208 (1991); <u>Fasselin v. BLM</u>, 102 IBLA 9, 14 (1988); and <u>Webster v. BLM</u>, 97 IBLA 1, 3–4 (1987). As will be shown below, many parts of the BLM's October 30, 2006 with regard to the Owyhee allotment decision are arbitrary and capricious and should be reversed.

C. The October 30, 2006 Decision with Regard to the Owyhee Allotment Is Arbitrary, Capricious and Inequitable

1. Under the October 30, 2006 ROD, grazing on the Owhyee allotment will be authorized on a two-year rotation system. The ROD allows 23,247 animal unit months ("AUMs") of use or 2300 head of livestock to rotate among certain pastures on the allotment on year one and 20,118 AUMs or 2150 head to rotate among certain pastures on year two; however, this grazing is authorized only after specific range improvements are constructed. These improvements are the South Fourmile Owhyee River Riparian Fence, the Fourmile Butte Well and the Fourmile Butte Well Pipeline and Troughs. Importantly, with regard to the timing for completion of the projects, the BLM states the construction of these projects is dependent upon the availability of funds. ROD at 24. There are no provisions in the ROD discussing how grazing will be authorized if funds are not available and the improvements are not built. Thus, Ellison does not know if it will even be allowed to graze its allotment in 2007.

The failure to provide for grazing on the Owhyee allotment if these improvements are not constructed equates to a failure to comply with BLM statutory mandates. The Owhyee allotment is a grazing allotment within a grazing district. The creation of a grazing district means that grazing must occur on the land. As summarized by the Tenth Circuit, "Congress intended that once the Secretary established a grazing district under the Taylor Grazing Act ("TGA"), the primary use of that land should be grazing." Public Lands Council v. Babbitt ("PLC 2"), 167 F.3d 1287, 1308 (10thCir. 1999). Thus, except upon the showing of good cause, the Secretary does not have discretion to bar grazing within a grazing district. See 43 C.F.R. § 4110.3 (1997) (stating that changes in grazing use "must be supported by monitoring, field observations, ecological site inventory or other data acceptable to the authorized officer.") The BLM has not presented any data sufficient to bar grazing on the Owyhee allotment if these range improvements are not constructed.

2. The FEIS states that the Upper and Lower Fourmile pastures will "possibly" be closed for fire rehabilitation in 2007. The FEIS also states that if these or other pastures are closed due to fire, livestock are to be <u>removed</u> from the public lands. <u>See</u> FEIS at 2-37. Again, these decisions are not consistent with BLM's regulations. While the BLM has the authority to close certain areas to grazing due to drought, fire or other emergency situations, the BLM is still require

The Secretary can modify the boundaries of a grazing district, but unless land is removed from designation as grazing, the Secretary must use it for grazing. 43 U.S.C. § 315. See generally, Mountain States Legal Foundation v. Andrus, 499 F.Supp. 383 (D. Wyo. 1980) (holding that the intent of Federal Land Policy Management Act ("FLPMA") was to limit the ability of the Secretary of the Interior to remove large tracts of public land from the operation of the public land laws).

to justify its closure decisions with data. Specifically, the BLM must establish by a preponderance of the evidence that Ellison's livestock grazing was causing an "imminent likelihood of significant resource damage" prior to suspending livestock grazing without more notice. 43 C.F.R. § 4110.3-3(b). The BLM cannot anticipate how large future fires will be, how much damage will occur and whether it needs to make contingency plans in emergency situations due to fire. Thus, the BLM's decision to automatically close pastures that have even been partially burned by fire or other emergency conditions fails to comply with agency regulations.

3. There are also numerous concerns with the ROD's requirements regarding its proposed monitoring and livestock removal requirements. First, both the FEIS and ROD state that if utilization in <u>any</u> key area is not met and livestock are determined to be the causal factor, grazing will be reduced.² According to the ROD, utilization will not be averaged across a use area or pasture. There is no indication in the ROD or the FEIS of the location of these key areas, what species will be monitored in the key areas and how the BLM will separate the substantial wild horse use of the key areas from the livestock use.

Second, the ROD does not state how utilization will be measured by the BLM. If the BLM is going to force Ellison to comply with utilization limitations, the BLM must accurately and scientifically measure those limits.

Third, in contrast to the ROD, the FEIS states that wild horse utilization is added to Ellison's livestock use. FEIS at 2-41. Once the maximum utilization is reached, whether it be reached by wild horses or livestock, livestock will be reduced.³ It is clearly arbitrary and capricious to force Ellison to remove its livestock if wild horses, over which Ellison has no control, are causing utilization limits to be reached. The utilization standards in the FEIS are:

- 1. Maximum combined utilization by both livestock and wild horse will not exceed 50 percent of current years growth for those pasture within the Horse Management Area.
- 2. Maximum utilization of 50 percent of current year's growth on key herbaceous species by the end of the grazing season within both Lower and Upper Fourmile Pastures.
- 3. Maximum utilization of 60 percent of current year's growth on crested wheatgrass within the Winters Creek Seeding Pasture by the end of the grazing season.

In this area, the ROD is stricter than the FEIS. Under the FEIS, if utilization in a key area was exceeded, Ellison could move its livestock to another area. FEIS at 2-41.

Since the ROD did not specifically address this point in the ROD, it is not clear which document controls this important issue.

Fourth, the FEIS predicts that grazing use can be increased if "monitoring shows the ability of the vegetative resource to sustain additional grazing use." FEIS at 2-38. However, the FEIS does not define what "sustain additional grazing use" means, nor does it state what type and how long the monitoring will be. Again, Ellison needs to know what type of monitoring will occur and over how long a period so that it can manage its livestock use to achieve an increase in AUMs.

The issues raised by the BLM's reliance on utilization to force livestock removal are substantial. For example, the BLM's use of utilization alone is not in accordance with standard rangeland practice. According to BLM's technical manuals, the BLM is supposed to conduct grazing utilization studies in accordance with the Utilization Studies and Residual Measurements, Interagency Technical Reference 1996. This technical reference contains explanations about how to set up sampling plots, when to use transects and when to use other methods. It also provides formulas and clear directions about how to conduct utilization sampling. If the BLM simply uses ocular estimates in conducting utilization monitoring, Ellison will likely be forced to reduce its grazing under the ROD. See ROD at 22 (stating "if, following an increase in AUMs, the short-term objectives are again exceeded, if short-term objectives are not being achieved during the initial cycle and livestock are determined to be the causal factor, use will be reduced. The reduction will be equal to the change suggested for the carrying capacity calculations for the specific key area where objectives are not being met (i.e., key area utilization will not be averaged over the entire Use Area or pasture")).

Additionally, utilization studies alone are not a valid method for monitoring overall rangeland health. Rather the BLM should also include actual use, climatic data, trend and other studies in determining rangeland health. In fact, the IBLA has rejected utilization studies because they did not adequately take into account the variability of the range in terms of forage production, vegetation, soils, and utilization. See Riddle Ranches Inc. v. BLM, 138 IBLA 82, 96 (1997). Nowhere in the ROD does it state that the BLM will take into account the individual features and attributes of the rangeland and seep resources in assessing the health of the allotment. Warnings to managers that utilization or stubble height should be used as tools and not as management objectives have been published by a number of range scientists. BLM range conservationists have described misuses of utilization by stating "and that's the story on utilization; a fine old range management tool with valuable but limited application, is now being used less as a management tool and more as a political tool for removing livestock and wild horses from rangelands." However, according to the ROD and FEIS, utilization monitoring is the only tool the BLM will consider in determining pasture use. This is not a valid assessment.

There is also no indication that the BLM's key areas will account for rangeland variability within the allotment. As stated above, neither the FEIS nor ROD describe the location of the key areas. Widespread use of the same set of grazing standards over large areas cannot be considered a reliable basis for decision making, because of the natural variation occurring within and between areas. Placing blanket utilization standards on an allotment and requiring livestock removal

based solely on reaching one standard in one key area is not informed or efficient management. If such standards are needed, they should be implemented on a *site-specific basis*.

4. As amended by sections seven and eight of the Public Rangelands Improvement Act of 1978, Section 402 of the Federal Land Policy and Management Act of 1976, 43 U.S.C. § 1752(d), requires consultation, cooperation and coordination with the impacted permittee prior to making decisions regarding his use of his allotment. See 43 C.F.R. § 4110.3-3. Additionally, the regulations state:

When the authorized officer determines that the soil, vegetation, or other resources on the public lands require immediate protection because of condition such as drought, fire, flood, insect infestation, or when continued grazing use poses an imminent likelihood of significant resource damage, after consultation with, or a reasonable attempt to consult with, affected permittees or lessees, the interested public, and the State having lands or responsible for managing resources within the area, the authorized officer shall close allotments or portions of allotments to grazing by any kind of livestock or modify authorized grazing use

43 C.F.R. § 4110.3-3(b) (emphasis added).

The Bureau's grazing regulations define "consultation, cooperation, and coordination" as an "interaction for the purpose of obtaining advice, or exchanging opinions on issues, plans, or management actions." See 43 C.F.R. § 4100.0-5. Specifically, by taking into account the livestock operator, the requirement of cooperation between the leasing agencies and the rancher allows for a directed and coordinated effort of rangeland improvement. See 124 Cong. Rec. H19507 (daily ed. June 29, 1978) (statement of Rep. Marlenee). The "consultation process" is not intended to be painstakingly difficult to administer, but requires several conversations between the interested parties and time to assess, study, plan and contemplate new alternatives and/or ideas with regard to the proper management of the public's rangeland resources. See M.L. Investment Co. v. Bureau of Land Management, 130 IBLA 376, 391–92 (1994). The regulations define consultation as, "interaction for the purpose of obtaining advice, or exchanging opinions on issues, plans, or management actions." 43 C.F.R. § 4110.0-5.

In this case, at no time did the BLM "consult, cooperate or coordinate" with Ellison prior to issuing this ROD. For example, the BLM failed to provide Ellison with a copy of the FEIS, nor notify Ellison that the FEIS was completed and published prior to issuing the ROD. The BLM did not consult with Ellison regarding the rotation system imposed by the ROD, nor about the decision to eliminate grazing from the allotment for fire, regardless of the timing, size or intensity of the fire. Disregarding the duty to consult, cooperate and coordinate with the permittee invalidates the agency's determination. See e.g. John L. Falen, 149 IBLA 347 (1999).

5. The APA governs the BLM's actions with regard to Ellison. The APA provides:

Except in cases of willfulness or those in which public health, interest, or safety requires otherwise, the withdrawal, suspension, revocation, or annulment of a license is lawful only if, before the institution of agency proceedings therefor, the licensee has been given —

- (1) notice by the agency in writing of the facts or conduct which may warrant the action; and
- (2) opportunity to demonstrate or achieve compliance with all lawful requirements.

5 U.S.C. § 558(c). The APA defines a license as "includ[ing] the whole or part of an agency permit, certificate, approval, registration, charter, membership, statutory exemption or other form of permission." 5 U.S.C. § 551(8). Section 558(c) of the APA applies to a grazing permit, such as the one held by Ellison. William J. Thoman, 157 IBLA 95, 103-104 (2002).

As stated above, the APA requires written notice and an opportunity to demonstrate or achieve compliance, all "before the institution of agency proceedings." In this instance, the ROD does not allow the BLM to follow the statutorily-mandated procedures. Under the ROD and FEIS, livestock grazing will be removed once utilization standards in a single key area are met, regardless of the use in the rest of the pasture. Livestock will also be removed if there is a fire, regardless of the size, intensity or location of the fire. Under these two scenarios, there is no opportunity for Ellison to show that it can continue to graze in the pasture. Under certain factual situations, either reaching a utilization standard or certain types of fire are not cases where there is a concern with "willfulness or those in which public health, interest, or safety requires otherwise" and, thus, suspension of Ellison's grazing may be inappropriate. Rather Ellison is entitled to written notice that would afford it the opportunity to show how continued grazing could occur despite the BLM's alleged "imminent likelihood of significant resource damage." See Air North America v. D.O.T., 937 F.2d 1427, 1438 (9th Cir. 1991) ("the purpose of section 558(c) is to provide individuals with an opportunity to correct their transgressions before the termination or suspension of their licenses.") (internal citations omitted).

In order for the BLM's action to suspend Ellison's grazing privileges to be valid, the BLM would have had to inform the permittee in writing about its concern about resource damage prior to issuing its decision. Then, Ellison would have the opportunity to respond with his range expert's data and coordinate with the BLM to determine how grazing could continue on the pasture with an alternate grazing plan. Without this opportunity for notice and a chance to respond, the suspension of Ellison's grazing privileges cannot be valid. See Capital Produce Co., Inc. v. United States, 930 F.2d 1077, 1079-81 (4th Cir. 1991) (holding that failure to provide prior written warning that conduct was deficient and an opportunity to correct deficiencies required that license suspension must be set aside); see also Hutto Stockyard, Inc. v. USDA, 903 F.2d 299, 304-05 (4th Cir. 1990) (same).

III. CONCLUSION

In conclusion, the decisions in the ROD as they relate to the Owhyee allotment are arbitrary, capricious and do not comply with the BLM's regulations. As such, the portion of the ROD pertaining to the Owhyee allotment should be reversed.

RESPECTFULLY SUBMITTED this 28th day of November, 2006.

Karen Budd-Falen

BUDD-FALEN LAW OFFICES LLC

300 East 18th Street

P.O. Box 346

Cheyenne, WY 82003-0346

(307) 632-5105 - Telephone

(307) 637-3891 - Facsimile

karen@buddfalen.com

CERTIFICATE OF SERVICE

I, the undersigned, hereby certify that a true and correct copy of the foregoing was deposited in the United States mail, certified/return receipt, on this 28th day of November, 2006, to the following:

7005 1820 0005 1449 2329

Tom Myers Friends of Nevada Wildlife 200 Bartlett Street Reno, NV 89512

7005 1820 0005 1449 2343

Hawkwatch International, Inc. 1800 S. West Temple, Suite 226 Salt Lake City, UT 84115

7005 1820 0005 1449 2367

Charles Watson Nevada Outdoor Recreation Association P.O. Box 1245 Carson City, NV 89701

7005 1820 0005 1449 2381

Marjorie Sill Sierra Club - Toiyabe Chapter 720 Brookfield Drive Reno, NV 89503

7005 1820 0005 1449 2404

Sara Barth
The Wilderness Society
P.O. Box 29241
San Francisco, CA 94129-0241

7005 1820 0005 1449 2220

Harvey Barns N1 Grazing Board HC 30 Box 347 Elko, NV 89801

7005 1820 0005 1449 2336

Friends of Nevada Wilderness P.O. Box 9754 Reno, NV 89507

7005 1820 0005 1449 2350

Sierra Club 408 C Street Washington, DC 20002

7005 1820 0005 1449 2374

Committee for the High Desert and Western Watersheds Project P.O. Box 2863
Boise, ID 83701

7005 1820 0005 1449 2312

Rose Strickland Sierra Club - Toiyabe Chapter P.O. Box 8096 Reno, NV 89507

7005 1820 0005 1449 2213

Grant Gerber Wilderness Impact Research Foundation 491 - 4th Street Elko, NV 89801

7005 1820 0005 1449 2237

John E. Hiatt Red Rock Audubon Society P.O. Box 96691 Las Vegas, NV 89193-6691

Paul Bottari Nevada High Country Tours P.O. Box 135 Wells, NV 89835

7005 1820 0005 1449 2268

Joe Guild Nevada Cattlemens Association P.O. Box 310 Elko, NV 89803

7005 1820 0005 1449 2282

Ronald P. McRobbie Air Force Regional Environmental Office 333 Market Street, Suite 635 San Francisco, CA 94105-2196

7005 1820 0005 1449 2305

Holland & Hart, LLP P.O. Box 2527 Boise, ID 83701

7005 1820 0005 1449 2121

Bill Hall Ellison Ranching Company HC 32 Box 240 Tuscarora, NV 89834

7005 1820 0005 1449 2145

Varlin Higbee P.O. Box 354 Alamo, NV 89001

7005 1820 0005 1449 2169

Elko Board of County Commissioners 569 Court Street Elko, NV 89801

7005 1820 0005 1449 2251

Robert McGinty 3154 Midland Drive Elko, NV 89801

7005 1820 0005 1449 2275

Nevada Department of Wildlife 60 Youth Center Road Elko, NV 89801

7005 1820 0005 1449 2299

National Headquarters National Mustang Association, Inc. P.O. Box 1367 Cedar City, UT 84721-1367

7005 1820 0005 1449 2114

Robert D. Williams U.S. Fish & Wildlife Service 1340 Financial Boulevard, Suite 234 Reno, NV 89502-7147

7005 1820 0005 1449 2138

John McLain Resource Concepts, Inc. 340 N. Minnesota Street Carson City, NV 89703-4152

7005 1820 0005 1449 2152

Nevada State Clearinghouse Department of Admin. 209 E. Musser Street, #200 Reno, NV 89701-4298

7005 1820 0005 1449 2176

Jim Andrae Doby George, LLC HC 32 Box 370 Tuscarora, NV 89834

Bernie Carter Doby George, LLC 165 W. Liberty, Suite 100 Reno, NV 89501

7005 1820 0005 1449 2206

Charles and John Young 5010 N. Hwy 38 Brigham City, UT 84302

7005 1820 0005 1449 2022

Jerry Goodwin Parasol Ranching, LLC P.O. Box 3707 Ogden, UT 84409

7005 1820 0005 1449 2046

Newmont Gold Company P.O. Box 669 Carlin, NV 89822

7005 1820 0005 1449 2060

Nevada State Office Bureau of Land Management P.O. Box 12000 Reno, NV 89520

7005 1820 0005 1449 2084

Scott Egbert HC 60 Box 135 Wells, NV 89835

7005 1820 0005 1449 2107

Steve Foree Nevada Department of Wildlife 60 Youth Center Road Elko, NV 89801

7005 1820 0005 1449 2190

Nevada Land & Resource Company 3480 GS Richards Boulevard Suite 101 Carson City, NV 89703

7005 1820 0005 1449 2015

Vidler Water Company 3480 GS Richards Boulevard Suite 101 Carson City, NV 89703

7005 1820 0005 1449 2039

Martha P. Hoots P.O. Box 36 Deeth, NV 89823

7005 1820 0005 1449 2053

Commission for the Preservation of Wild Horses 885 E. Lake Boulevard Carson City, NV 89704

7005 1820 0005 1449 2077

Big Springs Ranch, LLC The Star Living Trust P.O. Box 81624 Las Vegas, NV 89118

7005 1820 0005 1449 2091

Noel Owens Independence Valley Ranch, LLC P.O. Box 463 Wells, NV 89835

7005 1820 0005 1449 1919

Bank of Utah 25 North Tremont Street Tremonton, UT 84337

Sherie R. Goring 9940 North Hwy 37 Deweyville, UT 84309

7005 1820 0005 1449 1940

Steven R & Daniel Chournos 1670 East 600 North Tremonton, UT 84337

7005 1820 0005 1449 1964

Jeffrey O. Roche 19205 North 6000 West Garland, UT 84312

7005 1820 0005 1449 1896

Kay Richins H & R Livestock P.O. Box 188 Henefer, UT 84033-0188

7005 1820 0005 1449 1971

Darrel Kippens and Sons P.O. Box 1045 Morgan, UT 84050

7005 1820 0005 1449 1995

Dawn Lappin Wild Horse Organized Assistance P.O. Box 555 Reno, NV 89504

7005 1820 0005 1449 1711

The Honorable Jim Gibbons U.S. House of Representatives 100 Cannon Building Washington, DC 20515

7005 1820 0005 1449 1933

Pine Valley Sheep Ranch, Inc. P.O. Box 162 Hyrum, UT 84319

7005 1820 0005 1449 1957

Jon Marvel Western Watersheds Project Box 1770 Hailey, ID 83333

7005 1820 0005 1449 1889

David D. Morris HC 63 Box 0001 Park Valley, UT 84329

7005 1820 0005 1449 1902

Ely Field Office Bureau of Land Management HC 33 Box 33500 Ely, NV 89301-9402

7005 1820 0005 1449 1988

Barbara Flores Colorado Wild Horse & Burro Coalition 2406 15th Avenue Court Greeley, CO 80631

7005 1820 0005 1449 2008

Andrea Lococo Animal Welfare Institute P.O. Box 138 Bondurant, WY 82922

7005 1820 0005 1449 1797

The Honorable Jim Gibbons U.S. House of Representatives 491 - 4th Street Elko, NV 89801

The Honorable Harry Reid United States Senate 528 Hart Senate Office Building Washington, DC 20510

7005 1820 0005 1449 1735

The Honorable John Ensign United States Senate 364 Russell Senate Office Building Washington, DC 20510-2805

7005 1820 0005 1449 1742

Chournos, Inc. 590 West 1600 East Tremonton, UT 84337

7005 1820 0005 1449 1759

Don Henderson Nevada Division of Agriculture 350 Capitol Hill Avenue Reno, NV 89502

7005 1820 0005 1449 1766

Bobby McEnaney c/o NRDC 1200 New York Avenue, NW Washington, DC 20005

7005 1820 0005 1449 1773

Hugh Stevens - Chair Elko Band Council 1745 Silver Eagle Elko, NV 89801

7005 1820 0005 1449 1803

The Honorable Harry Reid United States Senate 400 S. Virginia Street, Suite 902 Reno, NV 89501

7005 1820 0005 1449 1612

The Honorable John Ensign United States Senate 600 East William, Suite 304 Carson City, NV 89701

7005 1820 0005 1449 1629

Lydia Garvey 429 South 24th Street Clinton, OK 73601

7005 1820 0005 1449 1636

District Ranger Mountain City Ranger District U.S. Forest Service 2035 Last Chance Road Elko, NV 89801

7005 1820 0005 1449 1643

Terrell D. Rich
Partners in Flight National Coordinator
U.S. Fish and Wildlife Service
1387 S. Vinnell Way
Boise, ID 83709

7005 1820 0005 1449 1650

Amos Murphy - Chair Confederated Tribes of the Goshute Reservation cc: Chrissandra Bullcreek P.O. Box 6104 Ibapah, UT 84034

Ronnie Woods - Chair South Fork Band Council 21 Lee Unit B 13 Spring Creek, NV 89815

7005 1820 0005 1449 1674

Chrissandra Bullcreek Confederated Tribes of the Goshute Reservatioin P.O. Box 6104 Ibapah, UT 84034

7005 1820 0005 1449 1681

Marcie Phillips
Duck Valley Sho-Pai Tribe
P.O. Box 219
Owyhee, NV 89832

7005 1820 0005 1449 1698

Kristi Begay - Chair Wells Band Council P.O. Box 809 Wells, NV 89835

7005 1820 0005 1449 1704

Jerry Milet - Chair Duckwater Shoshone Tribe P.O. Box 140068 Duckwater, NV 89314

7005 1820 0005 1449 1513

Hugh Stevens - Chair Temoak Tribal Council 525 Sunset Street Elko, NV 89801

7005 1820 0005 1449 1520

Lisa Cagle - Chair Yomba Shoshone Tribe HC 61 Box 6275 Austin, NV 89310

7005 1820 0005 1449 1667

Alfreda Jake Elko Band Council 1745 Silver Eagle Elko, NV 89801

7005 1820 0005 1449 1568

Terry Gibson - Chair Duck Valley Sho-Pai Tribe P.O. Box 219 Owyhee, NV 89832

7005 1820 0005 1449 1575

Sallas Smales South Fork Band Council 21 Lee Unit B 13 Spring Creek, NV 89815

7005 1820 0005 1449 1582

Aurora Aboite Wells Band Council P.O. Box 809 Wells, NV 89835

7005 1820 0005 1449 1599

Annette George Duckwater Shoshone Tribe P.O. Box 140068 Duckwater, NV 89314

7005 1820 0005 1449 1605

Jody Abe Temoak Tribal Council 525 Sunset Street Elko, NV 89801

7005 1820 0005 1449 1414

Bonnie Bobb Yomba Shoshone Tribe HC 61 Box 6275 Austin, NV 89310

Joseph Holley - Chair Battle Mountain Band Council 37 Mountain View Drive, Suite C Battle Mountain, NV 89820

7005 1820 0005 1449 1544

Reggie Premo Western Shoshone Committee P.O. Box 425 Owyhee, NV 89832

7005 1820 0005 1449 1551

Cindy Marques Ely Shoshone Tribe 16 Shoshone Circle Ely, NV 89301

7005 1820 0005 1449 1452

Andrew J. Orahoske 1991 Columbia Street Eugene, OR 97403

7005 1820 0005 1449 1469

Richard Artley 415 East North 2nd Grangeville, ID 83530

7005 1820 0005 1449 1476

Ginger Harmon Box 10191 Ketchum, ID 83340

7005 1820 0005 1449 1483

Wayne Y. Hoskisson 263 South 100 East Moab, UT 84532

7005 1820 0005 1449 1490

Craig C. Downer, Ecologist P.O. Box 456 Minden, NV 89423

7005 1820 0005 1449 1421

Gregory Holley
Battle Mountain Band Council
37 Mountain View Drive, Suite C
Battle Mountain, NV 89820

7005 1820 0005 1449 1438

Dianna Buckner - Chair Ely Shoshone Tribe 16 Shoshone Circle Ely, NV 89301

7005 1820 0005 1449 1315

Karen Klitz 1622 62nd Street Berkeley, CA 94703-1271

7005 1820 0005 1449 13146

Bob Brister 1102 South 800 East #A Salt Lake City, UT 84105

7005 1820 0005 1449 1353

Michael "Buffalo" Mazzetti P.O. Box 433 Tonasket, WA 98855

7005 1820 0005 1449 1360

Gail Fox 393 Jerseytown Road Bloomsburg, PA 17815

7005 1820 0005 1449 1377

B. Sachau 15 Elm Street Florham Park, NJ 07932

7005 1820 0005 1449 1384

Mary Jones 413 Brassy Street Lewiston, MT 59457

Veronica Egan Great Old Broads for Wilderness P.O. Box 2924 Durango, CO 81302

7005 1820 0005 1449 1315

Matthews and Wines, PC 687 Sixth Street, Suite 1 Elko, NV 89801

7005 1820 0005 1449 1322

Barb Adams 7716 East Hillgate Way Parker, CO 80134

7005 1820 0005 1449 1339

Attn: Gwen
Office of the Environmental Policy and
Compliance
U.S. Department of the Interior
1849 C Street, NW MS 2342
Washington, DC 20240

7005 1820 0005 1449 1834

Donna Fitzgerald United States Department of Justice Environmental & Natural Resource Division P.O. Box 663 Washington, DC 20044-0663

7005 1820 0005 1449 1841

Merlin McColm Elko County Conservation Association P.O. Box 5010 Elko, NV 89802

7005 1820 0005 1449 1391

Van Hyning & Associates, Inc. 6835 43 Street SW Great Falls, MT 59404

7005 1820 0005 1449 1407

Duane James
United States Environmental Agency
75 Hawthorne Street
San Francisco, CA 94105

7005 1820 0005 1449 1810

Todd Tucci Advocates for the West P.O. Box 1612 Boise, ID 83701

7005 1820 0005 1449 1827

Nancy Zahedi Office of the Regional Solicitor Pacific Southwest Region 2800 Cottage Way, E-1712 Sacramento, CA 95825

7005 1820 0005 1449 1865

Jim Christison Julian Creek Ranch HC 66 Box 9 Golconda, NV 89414

7005 1820 0005 1449 1872

Juliana Harris Enviroscientists 4600 Kietzke Lane, Suite C129 Reno, NV 89502

Eric Lawton U.S. Department of the Interior Bureau of Land Management 1620 L Street NW, Room 204 Washington, DC 20036

Vancy E. Carlis e

FINAL GRAZING MANAGEMENT DECISION

AND

RECORD OF DECISION

FOR THE

SHEEP COMPLEX, BIG SPRINGS AND OWYHEE GRAZING ALLOTMENTS

File Code: 1793/4130 (NV-012)	
EIS Name: Sensitive Bird Species En	vironmental Impact Statement
Date Final EIS filed with EPA: May 31	, 2006 (INT-FES-06-14)
Preparing Office:	U.S. Department of the Interior Bureau of Land Management Elko Field Office 3900 East Idaho Street Elko, Nevada 89801
Cooperating Agency:	Nevada Department of Wildlife
Location:	Elko County, Nevada
Approved by: /s/ Helen M. Hankins Manager, Elko Field Office	10/30/2006 Date

SUMMARY

On April 14, 2003, three final multiple use decisions (FMUDs) made by the Bureau of Land Management (BLM), Elko Field Office, for the Sheep Allotment Complex, Big Springs and Owyhee allotments were challenged in the United States District Court, Nevada. On August 18, 2004, the court ordered BLM to complete an environmental impact statement (EIS) to determine impacts of livestock grazing with respect to the following sensitive birds:

- Sheep Allotment Complex -- western burrowing owls, raptors and sage-grouse
- Big Springs Allotment -- sage-grouse
- Owyhee Allotment -- western burrowing owls, raptors and sage-grouse;.

The grazing allotments are all located in Elko County, Nevada. The Sheep Complex includes nine allotments located in the southeast portion of the Elko district. The Big Springs FMUD covers two allotments, north and west of the Sheep Complex. The Owyhee Allotment is in the northwest corner of the district. As a result of the EIS process, BLM is deciding to modify each of the FMUDs for grazing management as described in this Record of Decision.

The Nevada Department of Wildlife participated in preparation of the EIS as a cooperating agency. *The Sensitive Bird Species Draft EIS* was distributed for public comment in December 2005, and the Final EIS at end of May 2006. To concurrently meet requirements for issuing grazing decisions, the Elko Field Office also mailed the three Proposed Grazing Decisions with the Final EIS to the interested public for protest.

The purpose of the proposed action is to manage livestock grazing in the subject allotments to maintain and enhance productivity for all rangeland values, including habitat of the sensitive bird species. For each of the decisions, the EIS analyzes the effects of four alternative grazing systems to the sensitive species and their habitat, including uplands, springs and riparian areas.

- 1. Re-issue Grazing Permits at Historic Levels (No Action) Alternative 1 is to re-issue the grazing permits that were in place prior to issuance of the multiple use decisions. It is used in the EIS to compare the results of the following three action alternatives.
- 2. Implement the Multiple Use Decision as Modified (Proposed Action) -- Alternative 2 would implement the 2001 Sheep Complex and 2002 Big Springs and Owyhee multiple use decisions with modifications for grazing management. The modifications include phasing in increases in authorized grazing use from an initial stocking level up to the livestock carrying capacity based on the results of monitoring to meet allotment-specific objectives established for each decision. It also establishes several short term key area utilization objectives and changes to the terms and conditions of the grazing permits. It further proposes range improvement projects essential to implement the grazing plan.
- 3. Permit Grazing without Riparian Exclosures and Vegetation Treatments Alternative 3 eliminated consideration of range improvements proposed under the multiple use decisions, and modified the periods and conditions under which grazing would be permitted to achieve riparian objectives without these projects.
- 4. Adjust Grazing in Key Sensitive Species Habitats Alternative 4 also provides a strategy to address concerns about grazing impacts to use of habitat by the sensitive species, including the potential for range improvements to spread non-native invasive weeds. Projects were eliminated where possible, and grazing season of use was adjusted to address specific habitat objectives.

The environmentally preferred alternative is Alternative 4. Other management considerations BLM weighed in deciding to implement Alternative 2 are identified. This decision incorporates all practical measures to mitigate impacts, and monitoring of use to ensure resource objectives are met. The last section of this document provides detailed descriptions of three Final Grazing Management Decisions, with references to changes made to each FMUD and the rationale for the change.

TABLE OF CONTENTS

1	INTRODUCTION	1
1.1	DECISION	1
1.2	BACKGROUND	1
2	ALTERNATIVES	3
2.1	ALTERNATIVE 1 – RE-ISSUE GRAZING PERMITS AT HISTORIC LEVELS (NO ACTION)	4
2.2		
	ACTION)	
2.3		
	TREATMENTS	
2.4		
3	ENVIRONMENTALLY PREFERRED ALTERNATIVE	
3.1	SHEEP ALLOTMENT COMPLEX	
3.2		
3.3		
4	MANAGEMENT CONSIDERATIONS	
5	MITIGATION AND MONITORING	
6	PUBLIC INVOLVEMENT	
6.1	SCOPING AND DRAFT EIS REVIEW	
6.1	COMMENTS ON THE FEIS AND PROTESTS ON THE PROPOSED GRAZING DECISIONS	
7	GRAZING MANA GEMENT DECISIONS	10
7.1	SHEEP ALLOTMENT COMPLEX GRAZING DECISION	
7.2	BIG SPRINGS ALLOTMENTS GRAZING DECISION	
8.3	OWYHEE ALLOTMENT DECISION	
8.4	IMPLEMENTATION	24
	TABLES	
Table	1: Sheep Allotment Complex - Comparison of Alternatives (AUMs)	5
Table	2: Big Springs Allotments - Comparison of Alternatives	6
	3: Owyhee Allotment - Comparison of Alternatives	
	4: Sheep Allotment Complex	
Table	5: Leppy Hills Allotment Spring Use System (April 1 - April 30)	13
	7: Boone Springs Allotment Grazing Rotation for the Period March 1 to March 31	
	8: West Big Springs Allotment Grazing System	
Table !	9: East Big Springs Allotment Grazing System	19
	10: Essential Range Improvements for East and West Big Springs Allotments	
	11: Interim Grazing System, Interim AUMs, and FMUD Allocation of Carrying Capacity for Active	
	se by Livestock for the Owyhee Allotment	
AUIC	12. Mange Improvements Essential for Imprementation of the Final Grazing System	····

MAPS

Sheep Allotment Complex – Existing & Proposed Use Areas & Range Improvements (FEIS,Map 2-1) Big Springs Patures, Use Areas and Existing Range Improvements (FEIS Map 2-2) Big Springs Alternatives, Proposed Range Improvements/Pastures (FEIS Map 2-3) Owyhee Allotment Proposed & Existing Range Improvements, Pastures & Use Areas (FEIS Map 2-5)

1 INTRODUCTION

1.1 DECISION

This decision is recorded in accordance with regulations for implementing the National Environmental Policy Act (NEPA) found at 40 CFR parts 1500-1508¹ and Bureau of Land Management (BLM) grazing regulations found at 43 CFR part 4100², as amended in August 2006 and published in the *Federal Register* on July 12, 2006 (71 FR 29402).

In May 2006, the BLM, Elko Field Office, completed the Sheep Complex, Big Springs and Owyhee Grazing Allotments Sensitive Bird Species Final Environmental Impact Statement (EIS). Based on my review of the alternatives analyzed in the EIS and as the BLM authorized officer and Field Manager, it is my decision to modify and implement the following three final multiple use decisions (FMUDs) previously issued by the Elko Field Office for grazing management, as described and analyzed in the EIS as Alternative 2:

- 2001 Sheep Allotment Complex FMUD
- 2002 Big Springs Allotment FMUD
- 2002 Owyhee Allotment FMUD

As stated in section 1.2 of the EIS, the purpose of the proposed action is to manage livestock grazing in the subject allotments to maintain and enhance productivity for all rangeland values, including habitat of the sensitive bird species. The EIS describes and analyzes four alternatives in detail. This Record of Decision (ROD) identifies factors BLM considered in selecting Alternative 2, which is to implement the FMUDs as modified for grazing management.

Section 2 of this document briefly discusses the alternatives that BLM considered for each of areas. Section 3 identifies environmentally preferred alternative Alternative 4, to adjust grazing in key sensitive species habitats. Section 4 discusses other management considerations that factored into BLM's selection of Alternative 2 implementation. Section 5 identifies mitigation and monitoring commitments. Section 6 discusses public involvement throughout the EIS process.

BLM's decision to implement and monitor modifications to the FMUDs for grazing management is described in detail in section 7. No changes to the wildlife and wild horse management decisions from each FMUD are made, so these decisions remain in effect.

The three final grazing decisions are subject to appeal in accordance with the BLM Grazing Regulations at 43 CFR 4160.4. Pending resolution of any appeal(s), the changes to grazing management will become effective in May 2007 for the Sheep Allotment Complex and in March 2007 for the Big Springs and Owyhee allotments.

1.2 BACKGROUND

On April 14, 2003, three FMUDs previously issued by the Elko Field Office for the Sheep Allotment Complex (which includes 9 allotments), Big Springs Allotment (which was split into 2 allotments), and Owyhee Allotment were challenged in the United States District Court for the District of Nevada with respect to the requirements of the Federal Land Policy and Management Act of 1976 (FLPMA) and its implementing regulations including the Fundamentals of Rangeland Health (FRH), 43 CFR subpart 4180, and NEPA.

The three FMUDs are available upon request to the Elko Field Office.

On August 14, 2004, Honorable Judge Howard D. McKibben upheld BLM's decisions in all respects, except for one. He directed the BLM to complete an environmental impact statement

The Council on Environmental Quality NEPA regulations were published in the Federal Register on Nov. 28, 1978,(43 FR 55990), with revisions published on April 25, 1986 (51 FR 15625)

² The BLM Grazing Regulations were amended in August 2006 and published in the *Federal Register* on July 12, 2006 (71 FR 29402).

(EIS) on livestock grazing management with respect to certain sensitive avian species associated with the subject grazing allotments. This ruling left the challenged FMUDs in place. Per Judge McKibben's Minute Order, the EIS is intended to determine impacts of livestock grazing (including both sheep and cattle) with respect to the following sensitive birds:

- Sheep Allotment Complex: Western burrowing owls, raptors and sage-grouse;
- Owyhee Allotment: Western burrowing owls, raptors and sage-grouse; and
- Big Springs Allotment: sage-grouse.

As ordered, "To the extent applicable to these sensitive species the BLM shall evaluate the impacts of grazing, considering springs, seeps and riparian areas, uplands habitat and land use plans."

The Sheep Complex includes nine allotments located in the southeast portion of the district. The Big Springs FMUD covers two allotments, located north and west of the Sheep Complex. The Owyhee Allotment is located in the northwest portion of the Elko District (FEIS, Appendix A, Map 1-1).

The Nevada Department of Wildlife participated as a cooperating agency in preparation of the EIS.

The purpose of the proposed action is to manage livestock grazing in the subject allotments to maintain and enhance productivity for all rangeland values, including habitat of the sensitive bird species. The need for action is to adjust grazing management to make significant progress toward meeting the *Standards and Guidelines for Rangeland Health* for the Northeastern Great Basin Resource Advisory Council area and achieve the multiple use objectives established by the Elko or Wells Resource Management Plan (RMP) (FEIS, section 1.2).

A Notice of Intent to prepare the EIS was published in the *Federal Register* on December 17, 2004. This notice initiated a 30-day public scoping period.

All of the action alternatives analyzed in the EIS conform to the applicable land use plan, i.e., the 1985 Wells RMP, as amended, for the Sheep Allotment Complex and the Big Springs Allotments, and the 1987 Elko RMP, as amended, for the Owyhee Allotment. The three action alternatives include the grazing management plan that specify the period of use and total number of animal unit months (AUMs) of livestock grazing in each allotment. They also include allotment-specific plant community objectives and interdisciplinary monitoring to protect and enhance the other resources.

The Draft EIS was filed and distributed for public review and comment in December 2005 (DES 05-70). By the end of the public comment period, the Elko Field Office received 450 comments in 40 comment letters from 29 individuals, permittees, local government, and a variety of local organizations. The BLM considered the public comments on the Draft EIS, revised the EIS as warranted, and prepared the Final EIS. The Final EIS was filed and sent to those on the distribution list (FEIS, chapter 4) on May 31, 2006 (FES 06-14). BLM also included in the mailing of the FEIS a Proposed Grazing Management Decision³ for protest, in accordance with grazing management regulations (43 CFR 4160.2), to the interested public involved in the decision-making process. Subsequent to the mailing of the FEIS, some parties, who had not received the proposed decisions requested the decisions and/or submitted protests. These were accepted and considered along with any other protests received.

³ NEPA regulations provide that a record of decision may be integrated into any other record prepared by the agency (40 CFR 1505.2). The regulations further provide that, when agencies have a formally established appeal process, the decision may be made and recorded at the same time the EIS is published. Regulations for grazing management at 43 CFR 4160 establish protest procedures for a proposed decision, followed by an opportunity for appeal of the final decision. Thus, BLM provided for the protest period on the proposed grazing decision to run concurrent with publication of the FEIS. The 30-day appeal period for this final grazing decision will run concurrent with issuance of this ROD.

Since issuance of the FEIS and Proposed Grazing Decision in May 2006, substantial parts of the Owyhee Allotment were burned by the Winters, Amazon and Silver Lake fires. Also, a small portion of the West Big Springs Allotment burned (West Pequop Fire). Actions necessary to temporarily close the burned areas to livestock use along with other temporary adjustments to the terms and conditions of the grazing permits, as warranted, are being analyzed and will be implemented through decisions separate from this Final Grazing Management Decision. 4 The FMUDs will not be in effect until the closed are are re-opened to livestock use. The wildfire closure decisions will describe the temporary adjustments to livestock use that differ from this final decision.

The Final Grazing Management Decision and ROD are combined in this document.

2 ALTERNATIVES

BLM considered a total of nine alternatives during the preparation of the EIS. Five of the alternatives were eliminated from detailed analysis. They include: (1) no grazing; (2) grazing permits based on drought conditions with temporary non-renewable use for non-drought years; (3) restore non-native vegetation to native vegetation; (4) change type of livestock from sheep to cattle (Sheep Allotment Complex, only); and (5) limit grazing to levels below average actual use. These five alternatives and reasons for their elimination are described in Section 2.6 of the FEIS.

For each of the subject allotments, four alternatives are described and analyzed in the EIS.

Alternative 1, the No Action alternative, was included for comparative purposes only. Monitoring and allotment evaluations had already demonstrated that this alternative was not meeting rangeland health standards.

⁴ The amended BLM Grazing Regulations that became effective after the FEIS was issued (on August 12, 2006) recognize BLM's authority to respond as necessary to drought, fire and other resource conditions. (43 CFR 4110.3).

All three action alternatives propose a grazing strategy and issuance of a term (10-year) grazing permit with terms and conditions. The action alternatives (i.e., Alternatives 2, 3 and 4) included grazing use adjustments under a phased approach, based on monitoring, to include:

- Introduction of stocking levels, in terms of the number of animal unit months (AUMs) initially authorized that are below the calculated carrying capacity for the allotment. This is referred to as an interim system in the FEIS.
- Disclosure of all allotment-specific objectives from the FMUDs, and addition of short-term objectives as terms and conditions on grazing permits, where necessary as a result of the EIS process.

Under the action alternatives, the difference between the *initial stocking levels* and *active grazing preference* (referred to as "permitted use" in the FEIS) would be held in *suspension* until monitoring shows that the allotment-specific objectives are being met. The active grazing preference was derived from the carrying capacity that was calculated in the FMUDs for each allotment. The active grazing preference is expressed in AUMs that would annually be permitted in a given allotment.

Actual use under the historic permits (Alternative 1) and the livestock carrying capacity of the allotments for the Proposed Action (Alternative 2) had been calculated as part of the allotment evaluation and FMUD process. For all three action alternatives, development and modification of the grazing strategy for each allotment identification of pastures and/or use areas and periods of use that would improve specific conditions of concern within the allotments. The AUMs that would be specified in the term permit for each allotment were based on the number of carrying capacity AUMs that may be removed in a given allotment, in accordance with the specified periods of use and permit terms and conditions.

The three action alternatives also propose range improvement projects to implement the grazing strategy. Protection of riparian habitat within a given allotment is dependent, in part, on construction of riparian exclosures or pasture fences. Water developments, such as wells, or piping water from the springs to troughs outside of a riparian area, are also integral for implementation of the grazing strategies for some of the alternatives. A distinction was made between proposed range improvement projects needed to implement each alternative and those that would facilitate the alternative. Those that would facilitate the alternative were included as reasonably foreseeable future actions, as they would only be implemented on an as needed basis.

All the alternatives incorporate standard operating procedures (SOPs) for construction of projects (Appendix B of the FEIS). The SOPs include protection measures for various resources to minimize potential impacts of any proposed range improvement project. All alternatives also incorporate allotment-specific objectives (Appendix C of the FEIS).

2.1 ALTERNATIVE 1 – RE-ISSUE GRAZING PERMITS AT HISTORIC LEVELS (NO ACTION)

Alternative 1 would continue the grazing systems that were in place prior to the allotment evaluations that were completed in 2000. These evaluations identified the number of AUMs authorized by existing livestock grazing permits. The allotment evaluations determined that the grazing systems were not achieving rangeland health standards, thus adjustments were necessary. This conclusion led to development of and issuance of the three multiple use decisions. Because continuation of grazing as historically permitted would not meet BLM's need for action, this alternative is included for comparative purposes only.

2.2 ALTERNATIVE 2 – IMPLEMENT THE MULTIPLE USE DECISION AS MODIFIED (PROPOSED ACTION)

This alternative followed the grazing system outlined in the MUD for each of the subject allotments, with the modifications with respect to management of livestock grazing as noted

above⁵. Alternative 2 was identified as BLM's Proposed Action and Preferred Alternative in the FEIS, and is selected in this decision document.

2.3 ALTERNATIVE 3 - PERMIT GRAZING WITHOUT RIPARIAN EXCLOSURES AND VEGETATION TREATMENTS

This alternative grazing system was developed as a result of public scoping comments expressing concerns regarding potential impacts of riparian exclosures and vegetation treatments proposed under the Multiple Use Decisions. The grazing system was modified to achieve riparian objectives without these range improvements. The periods and conditions under which grazing would be permitted were adjusted under this alternative to accommodate the riparian protection goals.

2.4 ALTERNATIVE 4 - ADJUST GRAZING IN KEY SENSITIVE SPECIES HABITATS

This grazing system was developed as a result of public scoping comments expressing concern about grazing impacts to use of habitat by the sensitive species, including the potential for range improvements to increase the establishment and spread of non-native species.

Under this alternative, range improvement projects were eliminated, as practicable, and grazing season of use was adjusted to address specific wildlife habitat objectives.

3 ENVIRONMENTALLY PREFERRED ALTERNATIVE

Identification of the environmentally preferable alternative involves a balancing of resource uses with that of resource protection. The environmentally preferred alternative best promotes our nation's policy as stated in section 101 of NEPA. It is the alternative that causes the least damage to the biological and physical environment and best protects, preserves and enhances the resources that are present.

⁵ The modifications to the FMUDs were developed as a result of scoping and public comment on the Draft EIS.

3.1 SHEEP ALLOTMENT COMPLEX

The three action alternatives for the Sheep Complex are compared with respect to the No Action Alternative (Alternative 1) in **Table 1**. Alternative 4 is the environmentally preferred alternative.

Alternative 1 is not proposed for implementation because it would not allow BLM to meet the rangeland health standards and manage livestock grazing in the subject allotments to maintain and enhance productivity for all rangeland values, including habitat of the sensitive bird species. Potential impacts to the plant communities and habitat for the subject sensitive species that would continue to occur would be most pronounced for owls in all nine allotment, and sage grouse in the Boone Springs Allotment, due to continued impacts to the riparian vegetation

and nesting habitat. Alternative 2 would improve the overall plant health (shrubs and grasses), improving habitat for the raptor prey species and for sage grouse through changes in grazing management. The range improvements would protect riparian habitat from impacts due to wild horses, with concomitant benefits to owls in the allotments, and sage grouse in the Boone Springs Allotment. Other raptor species would also benefit by the improved riparian habitat as these areas would attract additional prey species. Non-native invasive weed species would continue to exist in the allotments. Some wildlife mortality due to exclosure fence collisions is possible; however, BLM fence construction standard operating procedures (SOPs) would reduce potential impacts and any impacts that do occur are likely to be offset by improved foraging conditions. the

Table 1: Sheep Allotment Complex - Comparison of Alternatives (AUMs)

	Alternative 1 No Action	Alternative 2 Proposed Action	Alternative 3 w/o Exclosures or Vegetation Treatments	Alternative 4 Adjust Grazing in Key Habitats
Grazing Preference (Alt 1) or Livestock Carrying Capacity (Alt 2-4)	39,915	26,652	Same as Alternative 2	Same as Alternative 2
Change From No Action (% Reduction)	n/a	-13,263 (or -33%)	Same as Alternative 2	Same as Alternative 2
Average Actual Use (Alt 1) Initial Stocking Rate (Alt 2-4)	17,573	17,474	Same as Alternative 2	Same as Alternative 2
Non-Use (Alt 1) or Suspended Use (% of Active Preference)	-22,342	-9,178 (or -34%)	Same as Alternative 2	Same as Alternative 2

Alternative 3 would also result in improvement of the upland vegetation. The riparian areas would benefit some due to attainment of appropriate management levels for wild horses, but heavy horse use at springs could continue due to the exclosures not being constructed. Similarly, impacts from non-native, invasive species would continue at reduced levels. The impacts to riparian areas would affect all the raptors, but especially the long-eared and shorteared owls. Sage grouse summer brood habitat, specifically the riparian areas in the Boone Springs Allotment, would remain degraded under this alternative. It would require more

time to meet the riparian habitat standards under this alternative as compared to Alternative 2. Alternative 4 would result in improvement in the vegetation similar to that anticipated under Alternative 2. Alternative 4 is preferred over Alternative 2 because the improvement could be realized sooner under this alternative than under other alternatives due to the elimination of grazing by sheep in April. Benefits to the raptor prey species are likely to be realized. Herding of sheep to avoid grazing of riparian areas until exclosures are built is expected to benefit shorteared owl, long-eared owl, sage grouse and the other raptors, similar to Alternative 2. Some mortality due to fence collisions is possible;

however, BLM fence construction SOPs would reduce potential impacts and any impacts that do occur are likely to be offset by the improved foraging conditions at the protected riparian areas.

3.2 BIG SPRINGS ALLOTMENTS

The three action alternatives for the West and East Big Springs allotments are compared with respect to the No Action Alternative (Alternative 1) in **Table 2**. Alternative 4 is the environmentally preferred alternative.

Alternative 1 is not proposed for implementation because it would not allow BLM to manage livestock grazing in the subject allotments to maintain and enhance productivity for all rangeland values, including habitat of the sensitive bird species. Continuing to permit grazing at historic levels under Alternative 1 would not meet the rangeland health standards. Impacts to riparian vegetation and sage grouse habitat would occur under this alternative as

riparian areas would continue to receive heavy use and water developments would continue to divert most or all of the water from several springs. If no action is taken, the potential would remain high for non-native, invasive species establishment.

Alternative 2 - Implement the Multiple Use Decision as Modified, would result in improvement of shrub and grass species. Some impacts may occur, but unlike Alternative 1, the rest built into the system between periods of use would allow the shrubs to recover and maintain vigor. Similarly, grasses would demonstrate some short-term impacts, but would receive rest to allow sufficient photosynthesis to maintain plant vigor. Some impacts would continue at the areas of concentrated used (i.e., at water developments) as these areas receive heavier use than the rest of the pastures in addition to the mechanical hoof disturbance. Riparian areas would benefit from grazing deferment and riparian pasture development.

Table 2: Big Springs Allotments - Comparison of Alternatives

	Alternative 1 No Action	Alternative 2 Proposed Action	Alternative 3 w/o Exclosures or Vegetation Treatment	Alternative 4 Adjust Grazing in Key Habitats
Grazing Preference (Alt 1) or Livestock Carrying Capacity (Alt 2-4)	21,983	16,963	15,808	14,509
Change From No Action (% Reduction)	n/a	-5,020 (or -23%)	-6,175 (or -28%)	-7,474 (or -34%)
Average Actual Use (Alt 1) Initial Stocking Rate (Alt 2-4)	13,581	13,601	13,601	14,509
Non-Use (Alt 1) or Suspended Use (% of Active Preference)	-8,402	-3,362 (or -20%)	-2,207 (or -14%)	0 (n/a)

The improvement in upland vegetation and riparian areas would decrease the potential for non-native, invasive species establishment. The construction of the allotment boundary and East Big Spring/North Pequop pasture fencess would create potential for these species to establish through surface disturbance that creates suitable seedbeds for non-native, invasive species. The net result, however, would be lower overall potential for non-native, invasive species.

The grazing system, spring exclosures, and riparian fencing would improve riparian habitat

as these areas would receive rest or protection during hot season grazing. The Proposed Action alternative would improve nesting habitat quality, reduce disturbance at leks, and improve summer brood habitat for sage grouse.

Alternative 3 would achieve the riparian goals without the riparian exclosures/fences, by designing the grazing system to benefit the riparian vegetation. The grazing system would improve upland grass and shrub vigor by providing sufficient rest between grazing periods. The potential for non-native, invasive

species to establish is less under Alternative 3 than Alternative 2. Fewer acres of surface disturbance would occur due to the elimination of fences. The riparian vegetation would also recover under this alternative, except where wild horses are the causal factor. The grazing system would provide deferred use or rest-rotation in all the pastures with riparian habitats. However, cattle would still have access to the riparian areas, so improvement would be less.

Alternative 3 would result in the improvement of some nesting habitat, but without the rehabilitation of the sagebrush in Holborn Pasture, the potential for nesting habitat in this area of degraded sagebrush would not be realized. Lek disturbance would be reduced.

The grazing system for Alternative 4 would result in improvement of the grass and shrub vigor. The rest or deferment of riparian vegetation as well as the upland vegetation, combined with reduced AUMs would provide for lower intensity of use and periods of rest for plants to recover from the effects of herbivory. The riparian improvement is anticipated to be more than Alternative 3, but not as much as Alternative 2.

Impacts from non-native, invasive species would be similar to Alternative 3. Sage grouse would benefit from this alternative due to improved nesting habitat and some improvement in summer brood habitat. Alternative 4 would result in the least amount of disturbance during sage grouse breeding, nesting, and brooding activities, so it is environmentally preferred.

3.3 OWYHEE ALLOTMENT

The three action alternatives for the Owyhee Allotment are compared with respect to the No Action Alternative (Alternative 1) in Table 3. Alternative 4 is the environmentally preferred alternative.

Under Alternative 1, the upland areas were improving under the historic grazing system, but the riparian rangeland health objectives were not being met. The impacts to the riparian vegetation under this alternative reduce the quality of sage grouse brood habitat and habitat for the long-eared owl, short-eared owl. Habitat for many of the prey species on which the raptors depend would also be degraded under this alternative. Therefore, BLM does not consider Alternative 1 to be a viable alternative.

Under Alternative 2, impacts to sage grouse breeding and nesting activities would continue to occur. This is due to the alternating early season use in Star Ridge and Dry Creek pastures, where 11 of the 12 documented leks within the allotment are located. This impact would continue to be partially offset by cessation of grazing by June 30, and the alternate year of rest. Alternative 2 would improve the upland vegetation (with some areas of continued adverse impact), decrease the establishment and spread of non-native species as compared to alternative 1, improve riparian vegetation. In addition, this alternative would improve habitat for raptors and their prey.

Table 3: Owyhee Allotment - Comparison of Alternatives

	Alternative 1 No Action	Alternative 2 Proposed Action	Alternative 3 w/o Exclosures or Vegetation Treatments	Alternative 4 Adjust Grazing in Key Habitats
Grazing Preference (Alt 1) or Livestock Carrying Capacity) (Alt 2-4)	30,155	29,903	27,837	20,706
Change From No Action (% Reduction)	n/a	-252 (or < -1%)	-2,318 (or -8%)	-9,449 (or -31%)
Average Actual Use (Alt 1) Initial Stocking Rate (Alt 2-4)	18,862	23,247	Same as Alt.2	20,706
Non-Use (Alt 1) or Suspended Use (% of Active Preference)	-11,295	-6,656 (or -22%)	-4590 (or -16%)	0 (n/a)

4 MANAGEMENT CONSIDERATIONS

The purpose of NEPA and this action closely mirrors BLM's multiple use and sustained yield mandates under the Federal Land Policy and Management Act. Both seek to sustain the health, diversity, and productivity of the public lands for the use and enjoyment of present and future generations. Alternative 4 may be the environmentally preferable alternative in terms of making progress toward meeting habitat standards and objectives. However, this outcome must be balanced against consideration of factors affecting the use of the grazing allotments by the permittees. BLM prefers Alternative 2 over Alternatives 3 and 4 because it offers a more practical means to permit and monitor livestock use in the grazing allotments.

Alternative 4 for the Big Springs and Owyhee allotments would result in faster improvement of most resource conditions, in comparison to alternative 2 or 3, but no difference is expected for the Sheep Complex. However, not every resource would improve faster, nor would every resource necessarily improve to a higher state of quality under alternative 4. Furthermore, the cost of getting this more rapid improvement would be deeper reductions in livestock use which would result in adverse economic impacts to the permittees.

As the analysis indicates, the end result of either Alternative 2 or Alternative 4 is good habitat condition. The selected alternative, Alternative 2 provides a balance between those reasonable measures necessary to protect and improve the existing resource values and making use of the subject lands for grazing. Habitat conditions will improve. Therefore, Alternative 2 - Implement the Multiple Use Decision as Modified, is the alternative best able to comply with all applicable laws, regulations, policy, and agency direction.

The BLM acknowledges that the selected alternative represents a balanced compromise between various competing resource objectives so that the resources are utilized in the combination that best meets the present and

future needs of the American people. Alternative 2 represents a realistic management balance, given current and projected budgets, which is expected to move resource conditions in the allotments (and especially those areas needing improvement) towards the desired future conditions outlined in the goals and objectives for the subject allotments.

The selected alternative (Alternative 2) further acknowledges that controlled livestock grazing is a valid and legitimate use over most of the area, and is expected to be effective in limiting or excluding grazing to protect key sensitive species habitat.

The subject allotments were assessed for compliance with applicable Standards for Rangeland Health (BLM 1997) prior to completion of this ROD. The assessments were included in the FEIS by reference. The selected Alternative will meet the requirements of 43 CFR 4180 for the authorized officer to take appropriate action where livestock grazing is a significant factor for not meeting, or for not making significant progress toward meeting, a particular rangeland health standard.

5 MITIGATION AND MONITORING

Standard operating procedures, other resource protection measures common to all the alternatives, and allotment-specific terms and conditions in the grazing permits are incorporated into Alternative 2 and will be implemented and monitored as described in the FEIS (sections 2.2.1 and 2.2.2, Appendix B, Appendix C)

An additional mitigation measure to be taken, as discussed in the *Conservation/Mitigation Recommendations and Residual Impacts* section of the FEIS (Sheep Complex, 3.2.4.2; Big Springs; 3.3.4.21 Owyhee; 3.4.4.2) is to require that permittees annually inspect trough areas and riparian exclosures to detect undesirable weed species, and notify BLM of any need to treat noxious weeds and/or seed species suitable to the site and use levels.

BLM is not committing to normally require that permittees annually inspect and seed all areas

receiving heavy use with desired perennial grass species. This is because BLM lacks funding and staff to monitor the effectiveness of this requirement.

All practical means to avoid or reduce environmental harm are hereby adopted, and will be monitored and periodically evaluated in accordance with implementation of this decision. Monitoring of the short-term objectives and periodic allotment evaluations will be used to ensure that the grazing plan is being implemented and progress is being made towards meeting the short- and long-term goals and objectives. Grazing management will be adjusted based on this monitoring.

6 PUBLIC INVOLVEMENT

6.1 SCOPING AND DRAFT EIS REVIEW

A Notice of Intent (NOI) to prepare the EIS was published in the Federal Register on December 17, 2004. This NOI initiated the formal public scoping period and invited the public to provide scoping input for the EIS during a 30-day scoping period. The public scoping period ended on January 18, 2005.

On January 12, 2005 a public meeting was held in an open house format at the BLM field office in Elko, Nevada. Following consideration of public input from the public scoping period, a Draft EIS was prepared and distributed for public review and comment in December 2005.

NDOW cooperated with BLM and the contractor, SRK Consultants, Inc., in preparation of the EIS. Issues addressed as a result of scoping are summarized at the end of chapter 1 of the EIS.

The DEIS was distributed for public review and comment in December 2005. By the end of the 45-day public comment period, the Elko Field Office received 40 comment letters, in which approximately 450 comments were identified. Eleven of the letters, comprising 284 comments were from Western Watersheds Project, the plaintiff in the case that led to the court order to prepare this EIS. In addition, another seventeen letters were from individuals prompted by an online message that Western Watersheds Project

transmitted from their website on January 19, 2006. These letters included 122 comments, most of which were identical to nine points raised made by the message about the DEIS. The remaining letters were from environmental organizations (2), the Elko County Commissioners, Nevada State Clearinghouse, Environmental Protection Agency, Nevada Cattleman's Association, the permittees, and other local ranchers and individuals.

In response to the comments on the DEIS, BLM modified the alternatives and analysis as warranted, and prepared the FEIS. Notice of the availability of the FEIS was published in the Federal Register on May 31, 2006, and it was mailed to those on the EIS distribution list (see Chapter 4 of the FEIS) for comment.

6.1 COMMENTS ON THE FEIS AND PROTESTS ON THE PROPOSED GRAZING DECISIONS

At the same time that the FEIS was distributed for comment, BLM issued a Proposed Grazing Management Decision for protest in accordance with 43 CFR 4160.2. The Proposed Grazing Management Decision was included in the certified mailing of the Final EIS to all interested publics who had been involved in the allotment evaluation and the FMUD process, and any new parties who asked to be added to this mailing list during the EIS process. During the public review period for the FEIS and protest period for the three Proposed Grazing Decisions, the BLM received 12 letters. BLM organized the comments/protests and found that most of the comments were either addressed in the FEIS, repetition of comments submitted on the DEIS, or out of scope.

Of the twelve letters, four were submitted as protests of the Proposed Grazing Management Decision by the Western Watersheds Project. BLM found many of their comments had been made on the DEIS and most had been addressed by the FEIS, as explained by the responses to the comments (FEIS, Appendix D). For new comments, one pointed out information for the Big Springs grazing management strategy that had been omitted from the FEIS and proposed

grazing decision. This change is included in the final decision (Section 7.2 of this document).

The letter from the U.S. Environmental Protection Agency indicated that the FEIS adequately addressed all of their concerns related to the draft document. The Nevada State Clearing House indicated that the State had no further comments on the FEIS. Six individuals that commented reiterated points raised by the Western Watersheds Project.

Public comments on the FEIS and protests of the Proposed Grazing Management Decisions are available for viewing in the Elko Field Office during normal business hours.

In addition to the comment letters received during the comment/protest period, one letter from Western Watersheds regarding fires was received on August 21, 2006. This letter expressed concerns regarding the large fire (Winters Fire; 238,462 acres) which had occurred in the area of the Owyhee Allotment as of that date, and questioned how the area was going to be managed. In the time since this letter, two additional fires (Amazon Fire; 108,563 acres and Silver Lake Fire; 2,500 acres), occurred in the area of the Owyhee Allotment, The Winters Fire impacted 4,775

7 GRAZING MANAGEMENT DECISIONS

Having conducted an assessment of the compliance with the Standards for Rangeland Health, considered a range of reasonable alternatives in the May 2006 Sensitive Bird Species Final Environmental Impact Statement (FEIS), and all comments on the FEIS and protests on the Proposed Grazing Decisions, it is my decision, as the Field Manager of the BLM, Elko Field Office, to adopt and implement Alternative 2 for the Sheep Allotment Complex, Big Springs Allotments, and the Owyhee Allotment., as described herein and in the FEIS. My final decision is to modify the Final Multiple Use Decisions (FMUDs) for the Sheep Allotment Complex issued on October 25, 2001, the Big Springs Allotments issued on September 17, 2002, and Owyhee Allotment issued on April 19, 2002, as described herein. Implementation of each decision is contingent

acres (94%) of the Winters Creek Seeding, 12,873 acres (33%) of the Chimney Creek Field and 43,919 acres (25%) of the Dry Creek Field. The Silver Lake Fire affected 1% of the Dry Creek Field. The Amazon Fire affected 2,774 acres (7%) of the Chimney Creek Field.

Also, one fire (West Pequop Fire; 1,978 acres) occurred in the North Pequop Mountain Pasture of the West Big Springs Allotment. It affected 3.7% of the pasture

As was documented in the FEIS in regards to the 2005 Wilson Complex Fire and as a standard procedure, BLM manages wildfire occurrence with emergency stabilization and rehabilitation plans, monitoring and livestock closures. Interdisciplinary teams evaluate the burned area and determine resource needs for stabilization and longer term rehabilitation. Livestock are removed and grazing is temporarily suspended from the burned areas, and the grazing permits are modified with a full force and effect decision to reduce authorized use (AUMs). Once the burned areas have recovered, as established through monitoring to meet objectives of the stabilization and rehabilitation plan, livestock resume under the use may allotment management plan of record.

upon the resolution of any appeals in accordance with grazing regulations found at 43 CFR 4160.3.

All aspects of the decisions are in full compliance with the existing land use plans. Resource protection measures, grazing permit terms and conditions, and monitoring actions described in the FEIS as actions common to all alternatives (section 2.2), standard operating procedures (Appendix B) and allotment-specific objectives (Appendix C) are also included as part of each decision.

In implementing each decision, the Elko Field Office, will reissue livestock grazing permits for the allotments from the FMUDs to incorporate the modifications to the terms and conditions for use as described for Alternative 2 in the FEIS and summarized below. Essential range improvement projects will be constructed as priorities, funding and manpower allow.

The management decisions for wild horses and wildlife in the FMUDs remain in effect. Specifics modifications to each FMUD for grazing management are detailed below.

7.1 SHEEP ALLOTMENT COMPLEX GRAZING DECISION

The Final Multiple Use Decision (FMUD) for the Sheep Allotment Complex issued on October 25, 2001, is modified for grazing management as follows.

1. Establish the total number of AUMs of grazing preference (active and suspended use) for livestock in the Sheep Allotment Complex as shown in Table 4 and described below. This modifies pertinent portions of decision #2 found on pages 5 to 16 of the FMUD for the Sheep Allotment Complex.

Rather than implement the grazing systems using the calculated carrying capacity AUMs (26,652 livestock AUMs of active use) as provided for in the FMUD, the livestock grazing systems will be implemented using the average actual use AUMs (17,474 total AUMs) for the period 1987 through 1999 as displayed under the column titled "Revised Preference" in Table

4 below. This represents a 34 percent reduction in active use described in the FMUD. The difference between active use based on average actual use AUMs and active use provided in the FMUD (i.e., 9,178 AUMs) will be held in suspension until short-term objectives are met, proposed range improvements are constructed, and monitoring shows the ability of the vegetative resources to sustain additional grazing use (e.g. during drought), as these apply to each allotment.

The short-term objectives will also be used to determine the sustainability of existing use and non-attainment of these objectives will result in adjustment of authorized use downward with reduced AUMs going back into suspension. The short-term objectives are the utilization objectives described in the FMUD. For the West White Horse Allotment, where the average actual use was higher than the Post-Evaluation Carrying Capacity (active preference) established in the Sheep Allotment Complex FMUD, the authorized grazing level will be the Post-Carrying Capacity Evaluation preference) from the FMUD.

Table 4: Sheep Allotment Complex
FMUD Allocation of Carrying Capacity for Active Use by Livestock Compared to the Revised Allocation of
Active and Suspended Use for Livestock (AUMs)

Allotment	FMUD Carrying Capacity	Revised Preference		•
	Livestock Active Preference	Livestock Active Use – Average Use 1987 – 1999	Suspended Use	Total Preference (Active + Suspended)
Leppy Hills	3,351	2,257	1,094	3,351
UT/NV North	3,704	2,115	1,589	3,704
UT/NV South	2,646	1,690	956	2,646
Lead Hills	5,609	3,314	2,295	5,609
White Horse	3,916	2,154	1,762	3,916
West White	465	325	0	465
Horse		325 465 ¹		
Sugarloaf	2,001	1,979	. 22	2,001
Ferber Flat	2,013	1,498	515	2,013
Boone Springs	2,947	2,002	945	2,947
Total	26,652	17,474	9,178	26,652
Two Use Areas w	vill be used one out of three year	rs; 465 AUMs are included	in the total.	

The AUMs in suspension will be restored when short-term objectives are achieved in all years of a grazing rotation cycle (i.e., one to four years, depending on the allotment). Please note that the Final EIS stated the grazing rotation cycles were three to four years; however, the cycles range from one to four years, depending on the allotment. The AUMs will be restored at a rate to be determined by utilization goals and objectives, but not beyond the increases suggested by the FMUD carrying capacity calculations.

If, following the increase in AUMs, the short-term objectives are again exceeded, or if short-term objectives are not achieved during the initial cycle and livestock are determined to be the causal factor, use will be reduced equivalent to the change suggested by the carrying capacity calculations for the specific key area where objectives are not being met (i.e., the key area utilization values will not be averaged over the entire Use Area or pasture, but will be based on the key area where objective(s) are not being achieved). This calculation will be used to set the allowable AUMs for the entire pasture or Use Area.

Rationale: The 2001 FMUD authorized sheep use in the complex up to a total of 26,652 AUMs. This level of authorized use was based on carrying capacity calculations. The calculated capacity was 9,172 AUMs over the average actual use during the evaluation period. Grazing at average actual use levels during the evaluation period resulted in small portions of the allotments being grazed excessively. Although the changes to terms and conditions for livestock use described in the FMUD are anticipated to correct concerns over excessive use, we believe the livestock permittees should demonstrate that they can prevent excessive use prior to authorizing increases above average actual use and that additional use is available. The authorized use for the allotments will be limited to the average actual use to give the permittees the opportunity to demonstrate that they can meet short-term objectives at the average actual use level, and that additional use is available. before we consider increases above that level. If the permittee is unable to demonstrate that they can meet the short-term objectives, further

reductions may be warranted. Limiting authorized use to the average actual use level also takes into account the unavailability of AUMs within ¼-mile of riparian areas until the riparian areas achieve proper functioning condition of are otherwise fenced as described under new terms and conditions below.

2. Drought closure decisions were issued in 2003. Allotments or portions of allotments that continue to be closed as of the date of this decision due to drought will be managed with the following criteria upon reopening and the first five years after reopening: If the utilization objectives are exceeded by one or more utilization class in a Use Area or pasture, that Use Area or pasture will be rested the following year. This is a new term and condition for grazing use not found in the FMUD.

Rationale: The criteria applicable to livestock use upon reopening of areas closed due to drought are necessary to ensure the drought affected plants remain healthy and the shrubs have the opportunity to grow to their full stature.

3. Modify the management systems and/or seasons of use for the following allotments within the Sheep Allotment Complex as described below. This modifies pertinent portions of decision #3 for the Leppy Hills, UT/NV North and Boone Springs allotments, found on pages 16 - 18 and page 26 of the FMUD for the Sheep Allotment Complex. The grazing systems in the Lead Hills, White Horse, West White Horse, Sugarloaf, Ferber Flat, and UT/NV South allotments will remain as outlined in the October 25, 2001 FMUD.

Leppy Hills Allotment.

The allotment will be divided into three use areas (Use Areas A, B, and Morris Basin). Use Area A will be located from the Playa reservoirs south to the allotment boundary and west of BLM road #1050. Use Area B will be located north and east of the Goshute Mountains and Use Area A. Authorized use will be from November 1 to March 31, with the exception of the Morris Basin Use Area, as described below. See Map 2-1 from the

FEIS (copy attached) for the locations of use areas.

The modifications pertinent to the Leppy Hills Allotment include removing Morris Basin as one of the two use areas regularly scheduled in the spring use rest rotation and replacing it with Use Area B. Morris Basin will continue to be available for use in April as described below; however, it is not included for use in the regular schedule.

Use Area A and Use Area B will be used on a rest rotation schedule for the spring use period from April 1 through April 30, as indicated in Table 5 below. Use in Morris Basin during April, when authorized, will be used as a substitute for use in Area A or Area B during this time.

Table 5: Leppy Hills Allotment Spring Use System (April 1 - April 30)

Year	Use Area
1	В
2	A
3	Repeat cycle

The Morris Basin Use Area is located in the Goshute Mountains. Approximately 450 AUMs occur in this basin, and grazing will be authorized on an annual review basis for this Use Area. When authorized, use will be from November 1 to December 1 and from April 1 to April 30. Please note the FEIS described the authorized periods of use in the Morris Basin Use Area as November 1 to December 1 and March 1 to March 31; however it was intended that use, when authorized, be November 1 to December 1 and April 1 to April 30. Unless specifically authorized in writing, no grazing will be allowed in the Morris Basin Use Area.

Rationale: Incorporating Use Area B in the Leppy Hills Allotment into the rest rotation system with Use Area A and, when authorized, Morris Basin, for the spring use period (April 1 to April 30) provides flexibility to adapt to weather conditions such as heavy snow in the mountains, and opportunities to graze cheatgrass that has invaded portions of Use Area B. Grazing cheatgrass in the spring can suppress

competition with native species and reduce fine fuels to slow the spread of future wildfires.

Utah/Nevada North Allotment.

The allotment includes three use areas. Authorized use will be from November 1 to March 31, with the exception of the Morgan Basin Use Area as described below. The Oana corral is located in both Use Area A and B, and the permittee will be allowed to utilize the corrals each year for loading and handling. For the period April 1 to April 30 each year, the grazing system will rotate between Use Area A and Use Area B as indicated in Table 6 below. Use in Morgan Basin during April, when authorized, will be used as a substitute for use in Area A or Area B during this time. Approximately 976 AUMs occur in the Morgan Basin Use Area, but grazing will be authorized on a pre-use review basis for this use area. Unless specifically authorized in writing no grazing will be allowed in the Morgan Basin use Area. See Map 2-1 from the FEIS (copy attached) for location of use areas.

The modifications pertinent to the Utah/Nevada North Allotment include removing Morgan Basin as one of the three use areas regularly scheduled in the spring use rest rotation. Morgan Basin will continue to be available for use from November 1 to December 1 and from April 1 to April 30 as described below; however, it is not included for use in the regular schedule.

Table 6: UT/NV North Allotment Grazing System for the Period April 1 to April 30

Year	Use Area
1	В
2	A
3	Repeat cycle

Rationale: Morgan Basin may at times have too much snow during April therefore it is not considered an area amenable to the regular rotation; however, it will still be available as a substitute for one of the other use areas providing additional rest for the other use areas during April.

Boone Springs Allotment.

The grazing system in the FMUD described two use areas, Areas A and B. Use Area A will include the land north and west of Alternate Highway 93, with a capacity of 947 AUMs. Use Area B will include lands south and east of Alternate Highway 93, with a capacity of 2,000 AUMs. See Map 2-1 from the FEIS (copy attached) for the locations of use areas.

Each use area could have been grazed from 11/1 to 3/31 with the amount of authorized use limited to the AUMs allocated to each use area. The modified grazing system below now limits the use in Use Area A from March 1 to March 31 and authorizes use in this western portion of the allotment to only one year in three (Table 7).

Table 7: Boone Springs Allotment Grazing Rotation for the Period March 1 to March 31

Year	Use Area		
1	A		
2	В		
3	В		
4	Repeat cycle		

Use Area B will be used during fall and winter and spring, November 1 to March 31. Use Area A will be used from March 1 to March 31. Use in Use Area A will be limited to one in three years and, during the year of authorized use, the sheep will be herded so as to avoid entering the area within ¼ mile of known sage grouse leks (strutting grounds) to limit any possible impacts to historic sage grouse leks in the western portion of the Boone Springs Allotment.

When Use Area A is grazed, active preference will be up to 947 AUMs and when Use Area B is grazed, active preference will be up to 2,000 AUMs, with total use not to exceed average actual use of 2,002 AUMs until short-term objectives are met, proposed range improvements are constructed, and monitoring shows the ability of the vegetative resources to sustain additional grazing use.

Rationale: Use in Use Area A will be limited to March 1 through March 31 and authorized only one in three years to reduce impacts to historic sage grouse leks in the western portion of the Boone Springs Allotment.

4. Modify the following short-term objectives.

The short-term objectives are utilization objectives for all the allotments. This modifies pertinent portions of decision #4 found on page 27 of the FMUD for the Sheep Allotment Complex. The changes clarify the timeframe to which the objectives are applicable.

- Maximum utilization of 60 percent of previous year's growth on key herbaceous species by the end of the grazing season;
- Maximum utilization of 50 percent of previous year's growth on salt desert shrub or other key shrubs (such as black sagebrush), by the end of the grazing season; and,
- c. Maximum utilization of 30 percent on current year's growth on salt desert shrub and other key shrubs (such as black sagebrush), and 50 percent on key herbaceous species in spring use areas, based on use of current year's growth at the end of spring use.
- Add the following new and/or modified specific terms and conditions to the grazing permits.
 - a. The livestock permittee is expected to move their livestock so as to not exceed established utilization objectives for previous year's growth on fall and winter use areas, and established utilization objectives for current year's growth in spring use areas. This term and condition modifies pertinent portions of the FMUD found under decision #3 to clarify that it is the permittee's responsibility to move the livestock.
 - b. If BLM determines that objectives are being approached and will be exceeded before the scheduled livestock removal

- date, the permittee will be notified and will have five calendar days to remove livestock to other areas not yet having approached objective use levels, or remove livestock from the allotment. This is a new term and condition for grazing use not found in the FMUD.
- c. If objective use levels are exceeded, scheduled off dates will be adjusted the following year to March 31, and remain in effect until monitoring for one complete grazing rotation indicates incremental extensions or further reductions in period of use are warranted. This modifies the pertinent portion of decision #4 and adds this term and condition to decision #5 found on pages 27-28 of the FMUD for the *Sheep Allotment Complex. Under decision #4 in the FMUD, we stated that if utilization was exceeded in two consecutive years, the scheduled off date would be adjusted to 3/31. This decision states that the adjustment will be made if use levels are exceeded in any one year and remain in effect untilmonitoring indicates adjustments are appropriate. In addition, this term and condition is added as a new term and condition to decision #5 in the FMUD.
- d.. Drought closure decisions were issued in 2003. Allotments or portions of allotments that continue to be closed as of the date of this decision due to drought will be managed with the following criteria upon reopening and the first five years after reopening: If the utilization objectives are exceeded by one or more utilization class in a Use Area or pasture, that Use Area or pasture will be rested the following year. This is a new term and condition for grazing use not found in the FMUD.
- e. No livestock grazing will be permitted within ¼-mile of seep or springs or along riparian areas until riparian projects are constructed or until Proper Functioning Condition (PFC) of the riparian areas is attained. This is a new

- term and condition for grazing use not found in the FMUD. This term and condition is added to ensure that livestock do not add to the impacts on riparian areas.
- f. Sheep camps will only be located in areas approved by the authorized officer. This is a new term and condition for grazing use not found in the FMUD. This term and condition is added in order to limit the impacts from sheep camps.
- g. Sheep bedding areas will only be located in areas approved by the authorized officer. Sheep may not be bedded in the same location more than seven consecutive days before being moved. Once moved, the next bedding area may not be within ¼-mile of the last bedding area. This is a new term and condition for grazing use not found in the FMUD. This term and condition is added in order to limit the impacts from sheep bed grounds.
- h. The permittee will submit a grazing application to the Elko Field Office prior to the start of each grazing year describing planned use within each Use Area. Planned use will be reviewed in relation to active preference. This is a new term and condition for grazing use not found in the FMUD. This term and condition will facilitate communication between the BLM and permittees to better ensure compliance with terms and conditions.
- i. Actual use reports will be specific to sheep camp/bedding areas within use areas. This is a new term and condition for grazing use not found in the FMUD. Specific information on when use has occurred in relation to sheep camp/bedding areas will help to better evaluate cause and affect relationships.
- 6. The following Conservation and Mitigation measures will be applicable as described below. These are new measures not previously included in the FMUD.

- a. Re-assess existing water developments to allow some spring flow to maintain the spring brook and associated riparian vegetation.
- b. Seasonal restrictions for grazing within a quarter-mile of the leks in Boone Springs Allotment (Use Area A) during the sage grouse breeding period will mitigate potential disturbance of sage grouse at the leks. This restriction is included in the terms and conditions for authorized sheep use in the Boone Springs Allotment as described under decision #3 above.
- c. Provide for occasional short-term grazing or other shrub treatment within spring exclosures to keep the riparian area at least partially as a meadow complex for sage grouse broods. The livestock permittee must receive specific BLM approval prior to grazing livestock within any spring/riparian exclosure.
- Resource protection measures, grazing permit terms and conditions, and monitoring actions described in the FEIS on pages 2-1 and 2-2 under "Actions Common to All Alternatives" will apply to all actions, as applicable.

All other decision points not affected by the above remain as outlined in the October 25, 2001 FMUD.

7.2 BIG SPRINGS ALLOTMENTS GRAZING DECISION

The Final Multiple Use Decisions (FMUD) for the Big Springs Allotment issued on September 17, 2002 is modified as follows. Changes include establishment of an interim grazing system with active preference allocated between the pastures of the West Big Springs Allotment and the East Big Springs as discussed below for each allotment.

West Big Springs Allotment

See Map 2-2 from the FEIS (attached) for the location of pastures/use areas,

1. Establish the total number of AUMs of grazing preference (active and suspended use) for livestock in the West Big Springs Allotment during operation of the interim grazing systems as described below. This modification is an addition to decision #2 applicable to livestock use found on pages 3-5 of the FMUD for the Big Springs Allotment.

The initial level of authorized use under the interim grazing system will be 3,651 AUMs of active use (see Table 8). The additional 1,137 AUMs of current active preference will be held in suspension until all essential range improvements have been completed and monitoring of the allotment with respect short-term key area objectives that additional demonstrates use is warranted. The short-term key objectives will also be used to determine the sustainability of existing use, and nonattainment of these objectives will result in adjustment of authorized use downward with reduced AUMs going back into suspension.

Upon completion of the essential range improvements described below, the final grazing system will be implemented. Following implementation of the final grazing system, and after monitoring demonstrates continued achievement of short-term objectives under the interim stocking levels, suspended AUMs will be reinstated. Suspended AUMs will be restored at a rate to be determined by utilization goals and objectives, but not beyond the increases suggested by the FMUD carrying capacity calculations. The results of monitoring will determine whether an increase, decrease, or continuation of grazing at existing levels is necessary.

Table 8: West Big Springs Allotment Grazing System

	Livestock U	Jse (AUMs)	Period of Use	
West Big Springs Allotment Pasture and Use Area	FMUD Carrying Capacity (Total Preference) Initial Stocking Level † Active Use)		YEARS 1 YEARS 2 and 4	
	4,788	3,651		engembalahk.
North Pequop Mountain Pasture				
North Use Area	1,168	No more than	7/1 - 9/30	7/1 – 9/30
South Use Area		1,168	Rest	5/1 - 6/30
Holborn Pasture	550		5/1 - 6/30	Rest
Independence Valley Pasture 2	3,050	2,463	9/1 - 6/30	9/1 - 6/30
Fenced Federal Range (FFR)	20	20		w.

The AUMs credited to owned and leased private lands intermingled with public lands would be reduced by the same percentage as public land permitted use grazing preference. The AUMs placed in voluntary non-use would be reinstated as the range improvements are implemented and as standards and guidelines are met.

2. Replace the Interim Grazing System described in decision #3 of the FMUD (pages 8-11) with the following Interim Grazing System:

The revised interim grazing system for the West Big Springs Allotment is displayed in Table 8 followed by the narrative description by pasture. This table and accompanying descriptions replace the individual tables and descriptions for each pasture contained within the West Big Springs interim grazing system portion of the FMUD. Grazing use shall not exceed the rated carrying capacity calculations for each pasture specified in the FMUD, with total use not to exceed the initial level of 3,651 AUMs of authorized use.

Holborn Pasture. Between early May and end of June, livestock will be moved from the Independence Valley Pasture into the Holborn Pasture north of Interstate 80. The rest rotation plan will allow growing season use one year followed by a complete year of rest. The amount of time livestock remain in the pasture is dependent on available water for adequate distribution. In dry years, livestock will be

moved to the North Pequop Mountain pasture earlier than the planned turn out date.

North Pequop Mountain Pasture. In normal precipitation years the pasture will receive deferment from livestock grazing in the North Use Area. Movement into this Use Area in July will coincide with seed ripe or seed dissemination for most of the forage plants, resulting in deferment (i.e., growing season rest) each year.

The South Use Area will be rested one year and receive growing season use the following year, alternating with the Holborn Pasture.

This deferred rotation plan/rest rotation plan will require the cattle to be moved to the North Use Area each year from the Holborn Pasture and to the South Use Area in alternate years from the Independence Valley Pasture. The permittee will be responsible for monitoring livestock drift to the east side of this pasture, where the adjoining permittee grazes, and moving his livestock back to the west side in a timely manner. An important measure of the interim grazing system will be to remove livestock that drift into the East Squaw Creek and Upper Beacon Spring areas until the proposed allotment boundary fence is constructed.

² Use in the Independence Valley pasture will be rotated through use areas as described in the original FMUD. In accordance with the 2002 FMUD, 300 AUMs of the total allowed would not be authorized unless stockwater is hauled to the northwest portion of the valley or if a new water source is developed in this area.

There is sufficient topography in the western half of this pasture to normally prevent most livestock from drifting back to the south end of this pasture from the north. The permittee will be responsible for monitoring the south use area for any drift, with the livestock moved back to the north end in a timely manner.

Livestock may be trailed through pastures or use areas scheduled for rest or in dates outside permitted seasons of use only to reach pastures scheduled for use. No overnight stops in closed or rested pastures/use areas will be allowed on such trail movements.

East Big Springs Allotment

See Map 2-2 from the FEIS (attached) for the location of pastures/use areas.

3. Establish the total number of AUMs of grazing preference (active and suspended use) for livestock in the East Big Springs Allotment as described below. This modification is an addition to decision #2 applicable to livestock use found on pages 3-5 of the FMUD for the Big Springs Allotment.

The interim grazing system includes authorized use up to 10,150 AUMs of active use annually (see Table 9), with 2,025 AUMs of current active preference to be held in suspension until all essential range improvements have been completed and monitoring of the allotment with respect to short-term key area objectives demonstrates that additional use is warranted. The short-term key area objectives will also be used to determine the sustainability of existing use, and non-attainment of these objectives will result in adjustment of authorized use downward, with reduced AUMs going back into suspension.

Upon completion of essential range improvements described below, the final grazing system will be implemented. Following implementation of the final grazing system, and after monitoring demonstrates continued achievement of short-term objectives under the interim stocking levels, suspended AUMs will be reinstated, at a rate to be determined by

utilization goals and objectives, but not beyond the increases suggested by the FMUD carrying capacity calculations. The results of monitoring will determine whether an increase, decrease, or continuation of grazing at existing levels is necessary.

4. Replace the East Big Springs Allotment Interim Grazing System described in decision #3 of the FMUD (pages 13-18) with the following Interim Grazing System:

The revised interim grazing system is outlined in Table 9 below. This table and the accompanying description, by pasture, replace those contained in the existing FMUD. Grazing use shall not exceed the rated carrying capacity calculations for each pasture specified in the FMUD, with total use not to exceed the interim level of 10,150 AUMs of authorized use.

In years 2 and 4, livestock will leave Shafter Pasture by 1 March. The cattle will use East Pequop Bench from 1 March through 15 June, using the rotation system between use areas outlined in the FMUD. The North Bench use area in this pasture will be deferred until last in this pasture to minimize any potential impacts to sage grouse strutting and nesting. Starting in mid-June, the heard will be split, with some cattle going to Payne Basin and Six Mile/Long Canyon pastures. Starting on 1 May, some livestock will move into the North Pequop Mountain Pasture, where they will remain until the end of June or until utilization objectives along East Squaw Creek are met. On July 1, these cattle will be moved to Collar and Elbow, East Squaw Creek, Railroad, and Windmill pastures. During September and October, cattle will be moved into the private fields. Cattle will reenter the range around early October, with some cattle going to Collar and Elbow Pasture and the rest going to East Pequop Bench Pasture. Starting around 1 December, all livestock will be moved to Shafter for the winter.

Livestock may be trailed through pastures or use areas scheduled for rest or in dates outside permitted seasons of use only to reach pastures scheduled for use. No overnight stops in closed or rested

pastures/use areas will be allowed on such trail movements.

Table 9: East Big Springs Allotment Grazing System

	Livestock U	Jse (AUMs)	Period o	of Use
East Big Springs Allotment Pasture and Use Area	FMUD Carrying Capacity (Total Preference)	Initial Stocking Level (Active Use)	YEARS 1 and 3	YEARS 2 and 4
	12,175	10,150		
North Pequop Mountain Pasture ² North Use Area South Use Area	1,762	1,250	7/1 – 9/30 Rest	Rest 5/1 – 6/30
Payne Basin Pasture and Long Canyon/Six=Mile Pasture	756	37.5	7/1 - 9/14 6/16 - 8/30	6/16 -9/5 6/16 - 9/5
Railroad Field	255	*	Reserved Use	7/01 - 8/30
Windmill Field/Seeding Pasture	420	1 142	8/1 - 9/15	7/1 - 8/30
East Squaw Creek Pasture	330	No more than 8,508 aggregate AUMs.	6/20 - 6/30 9/1 - 10/20 (15 days during this period)	7/01 – 8/30
Collar and Elbow Pasture ³	1,899		6/16 – 1/31	7/1 - 12/15
East Pequop Bench Pasture 4	3,069	Carrying		
Shafter Pasture	3,396	Capacity will not be	11/1 - 4/15	12/1 - 3/1
North of Home Pasture	116	exceeded in	Drift Use	Drift Use
Squaw Creek Ranch Pasture	55	any pasture.	Drift Use/Gather	Drift Use/Gather
Lower Squaw Creek Ranch Pasture	100		Drift Use/Gather	Drift Use/Gather
Fenced Federal Range (FFR)	17	17	n/a	n/a

¹ The AUMs credited to owned and leased private lands intermingled with public lands would be reduced by the same percentage as public land permitted use grazing preference. The AUMs placed in voluntary non-use would be reinstated as the range improvements are implemented and as standards and guidelines are met.

Rationale: The interim grazing systems implemented in the FMUD assumed that the planned range improvements would be in place in a short amount of time. The existing interim

grazing systems expected an unrealistic level of livestock control without the aid of fences. The revised interim plan outlined above maintains the two years out of four deferment during the

² Livestock authorized in the North Pequop Mountain Pasture will be removed in a timely manner so that at the end of the growing season or grazing season, whichever occurs later:

a) a minimum of four inches average stubble height of selected key herbaceous riparian species (sedges/rushes) will be left along the stream banks of East Squaw Creek and;

b) Use on current year's growth of aspen and willow along East Squaw Creek is 35 percent or less.

³ Collar and Elbow Pasture will be open as a place to move cattle when utilization objectives on East Squaw Creek in the North Pequop Mountain Pasture are met.

⁴ Use of the East Pequop Bench Pasture will be rotated through use areas as described in the FMUD, with the exception of the North Bench area in East Pequop Bench, which will be deferred to last every year to minimize conflicts with sage grouse strutting and nesting.

growing season included in the original system. The East Squaw Creek watershed will be closed to grazing two years out of four, with a riparian friendly spring use period occurring during the years the pasture is grazed. Livestock will be present in the southern use areas of the North Pequop Mountain Pasture of both allotments during the same time every other spring; some mixing of cattle between the two sides will be expected, but this will eliminate the possibility of livestock drifting across the division line into rested use areas.

5.Construct the range improvements shown in Table 10 to transition grazing management from the Interim to the Final Systems. See Map 2-3 from the FEIS (attached) for the location of proposed projects. The Final Grazing System for the will be implemented when the allotment

boundary fence is completed and the essential projects in each pasture are completed. The BLM considers the improvements identified in Table 10 as essential to implementing the Final Grazing Plans outlined in the FMUD. The other improvements identified in the FMUD are analyzed in the FEIS as reasonably foreseeable future actions.

Rationale: This list represents the improvements that must be installed to allow the Final Grazing Plan to be implemented. The Revised Interim Grazing Plan will remain in place until these improvements are in place.

Table 10: Essential Range Improvements for East and West Big Springs Allotments

Project	Allotment/Location	Units
Allotment Boundary Fence West and East Big Springs/Pequop Summit		3 miles
Pasture Fence	East Big Springs/North Pequop Mountains East Squaw Creek	3 miles
Riparian Pasture Fence	East Big Springs/North Pequop Mountains	1½ miles
Exclosures and troughs	East Big Springs/North Pequop Mountains and Payne Basin	To Be Determined

6. Modify some of the short-term riparian objectives, as displayed in Appendix C of the Final EIS. Both the interim and final grazing systems will be governed by achievement of the short-term key area utilization objectives outlined in the FMUD, as modified.

Rationale: Modifications to some of the short-term riparian objectives were warranted based on what is practical to achieve under the interim and final grazing systems. The revisions pertain to the timeframes for achievement of upwards trends in functioning condition and proper functioning condition on riparian areas. For example, upward trends in functioning conditions are expected to occur in the southern portion of the North Pequop Mountain Pasture within the East Big Springs Allotment during the interim grazing system, with proper

functioning conditions to be achieved following installation of the essential fences and exclosures. Most of the riparian habitat is located in this pasture. Riparian areas located in other pastures are expected to improve and reach proper functioning condition following installation of exclosures.

- 7. Add the following Terms and Conditions to the Grazing Permits:
 - a. The livestock permittees are expected to move their livestock so as to not exceed established short-term key area objectives.
 - b. If BLM determines that objectives are being approached and will be exceeded before scheduled livestock removal date, permittee will be notified and will have five calendar days to remove livestock to other

areas within the pasture/use area not yet having approached objective use levels, to the next pasture in the schedule, or off the allotment.

- c. If short-term key area utilization objectives are exceeded, period of use for the next grazing period in that pasture/use area will be reduced by a minimum of two weeks where it shall remain until additional changes are indicated through monitoring. Period of use adjustments will apply to the next grazing season.
- d. Period of use extensions will be authorized only after two consecutive years of use with monitoring which indicates incremental extensions in period of use are warranted.

Rationale: These objectives will ensure progress is made towards achieving the Standards for Rangeland Health.

- 8. The following Conservation and Mitigation measures will be applicable as described below. These are new measures not previously included in the FMUD.
 - a. Provide for occasional short-term grazing or other shrub treatment within spring exclosures to keep the riparian area at least partially as a meadow complex for sage grouse broods. The livestock permittee must receive specific BLM approval prior to grazing livestock within any spring/riparian exclosure.
- Resource protection measures, grazing permit terms and conditions, and monitoring actions described in the FEIS on pages 2-1 and 2-2 under "Actions Common to All Alternatives" will apply to all actions, as applicable.

All other decision points in the Sept 17, 2002 FMUD not affected by the above modifications remain the same.

8.3 OWYHEE ALLOTMENT DECISION

The Final Multiple Use Decision (FMUD) for the Owyhee Allotment issued on April 19, 2002, is modified for grazing management as follows.

1. Establish the total number of AUMs of grazing preference (active and suspended use) for livestock in the Owyhee Allotment during operation of the interim grazing system as shown in Table 11 and described below. See Map 2-5 from the FEIS (attached) for the location of pastures/use areas. This modification is added to decision #1b found on pages 4-5 of the FMUD for the Owyhee Allotment.

Rather than implement the interim grazing system using the calculated carrying capacity (29,903 AUMs of active use in Year 1 (odd years) and 27,879 AUMs in Year 2 (even years) authorized in the FMUD, the initial stocking levels for the interim grazing system will be 23,247 AUMs in year 1 (20,118 in year 2) which is equivalent to the average actual use AUMs during the period 1995 through 2005. The difference between the calculated carrying capacity (i.e., 29,903 AUMs) and the average actual use AUMs (i.e., 23,247 AUMs), or 6,656 AUMs, will be held in suspension until short-term objectives are met, essential range improvements are constructed; monitoring shows the ability of the vegetative resource to sustain additional grazing use. The short-term key area objectives will also be used to determine the sustainability of existing use. Nonattainment of these objectives will result of adjustment authorized downward, with reduced AUMs going back into suspension. The terms and conditions and short-term key area objectives will apply to both the interim (initial) and final grazing systems.

Upon completion of the essential range improvements described below, the final grazing system will be implemented. Following implementation of the final grazing system, and after monitoring demonstrates continued achievement of short-term objectives after one cycle (i.e. two years) under the interim stocking levels, suspended AUMs will be reinstated by phasing-in the increase over a three-year period. Suspended AUMs will be restored at a rate to be determined by utilization goals and objectives, but not beyond the increases suggested by the FMUD carrying capacity calculations. The results of monitoring will determine whether an increase, decrease, or continuation of grazing at existing levels is necessary.

If, following an increase in AUMs, the short-term objectives are again exceeded, if short-term objectives are not achieved during the initial cycle and livestock are determined to be the causal factor, use will be reduced. The reduction will be equivalent to the change suggested by the carrying capacity calculations for the specific key area where objectives are not being met (i.e., the key area utilization values will not be averaged over the entire Use Area or pasture, but will be based on the key area where objective(s) are not being achieved) or other adjustments will be made to resolve the issue.

Table 11: Interim Grazing System, Interim AUMs, and FMUD Allocation of Carrying Capacity for Active
Use by Livestock for the Owyhee Allotment

Year	Pasture	Livestock Number ¹	Begin Period	End Period	Interim AUMs ²	FMUD Permitted AUMs
	Star Ridge	2,300	3/1	6/30	9,041	12,101
	Lower Fourmile	1,700	7/1	9/20	4,572	6,403
*	Upper Fourmile	600 ²	7/1	8/25	1,083	1,069
1.	Chimney Creek	600 2,300	8/26 9/21	9/20 11/30	503 <u>5,261</u> 5,764	7,543
	Dry Creek	Rest	Rest	Rest	Rest	Rest
	Winters Creek ³	variable		ing or Late	2,787	2,787
			-	Total	23,247	29,903
	Dry Creek	1,150 2,150	3/1 5/26	5/25 7/10	3,186 3,186 6,372	10,07,7
×į	Winters Creek	1,000	3/1	5/25	2,771	2,787
2	Chimney Creek	2,150	7/11	9/25	5,334	7,543
	Lower Fourmile	2,150	9/26	11/30	4,572	6,403
8 2	Star Ridge	Rest	Rest	Rest	Rest	Rest
	Upper Fourmile ³	variable		ing or Late	1,069	1,069
		6		Total	20,118	27,879

¹ - The final grazing system under Alternative 2 in the Final EIS shows 48 head of domestic horses from 3/1-12/15 for 444 AUMs. This use was combined with the cattle use for the interim system in order to keep a consistent sized herd of cattle. If the permittee wishes to use 444 AUMs with domestic horses, he may do so within the dates outlined for the pasture.

² – The interim AUMs for each pasture are based on grazing a consistent number of livestock on an annual basis. The difference between the initial level of authorized use under the interim grazing system and the FMUD allocation of carrying capacity will be held in suspension until essential range improvements have been completed and monitoring of the allotment with respect to short-term key area objectives demonstrates that additional use is warranted..

³The Winters Creek Pasture and Upper Fourmile Pasture will act as a "utility pasture" in alternate years to be used as needed. This might involve gathering, branding, weaning, etc. It may also be used as needed if water is limited in other pastures or if utilization objectives are close to being exceeded prior to the off date in other pastures. The carrying capacity for the pastures shall not be exceeded.

Rationale: The interim grazing systems (even and odd years) implemented in the Owyhee FMUD assumed that the planned range improvements would be in place in a short amount of time after the issuance of the FMUD. While the grazing systems outlined in the FMUD were making progress towards the attainment of the standards for rangeland health, this progress was occurring at a slow rate. Range improvement projects for the Lower Fourmile Pasture were proposed to make progress toward attainment of riparian objectives in this pasture. The interim system devised for the Owyhee Allotment maintains the two-pasture rest rotation between the Star Ridge and Dry Creek Pastures. It also maintains the growing season deferment in the Chimney Creek and Lower Fourmile Pastures, but it eliminates the early spring use in the Chimney Creek and Lower Fourmile Pasture. The interim system curtails livestock AUMs in the Lower Fourmile Pasture at average actual use. The limit of average actual

use within this pasture also limits use in the remaining five pastures within the allotment.

- Livestock may be trailed through pastures or use areas scheduled for rest or in dates outside permitted seasons of use only to reach pastures scheduled for use. No overnight stops in closed or rested pastures/use areas will be allowed on such trail movements.
- 3. Construct the range improvements shown in Table 12 to transition grazing management from the Interim to the Final System. See Map 2-5 from the FEIS (attached) for the location of proposed projects. The Final Grazing System will be implemented when these range improvements are completed.

Rationale: This list represents the improvements that must be installed to allow the Final Grazing Plan to be implemented. The Interim Grazing Plan will remain in place until these improvements are installed.

Table 12: Range Improvements Essential for Implementation of the Final Grazing System

Project	Pasture	Units	
South Fourmile Owyhee River Riparian Fence	Lower Fourmile	4 miles	
Fourmile Butte Well	Lower Fourmile	1 well	
Fourmile Butte Well Pipeline and Troughs	Lower Fourmile	9.5 miles	

4. Add the following new and/or modified specific terms and conditions to the grazing permit:

The livestock permittee is expected to move his livestock so as to not exceed established short-term key area objectives. This term and condition will help ensure progress is made towards achieving the Standards for Rangeland Health.

5. Resource protection measures, grazing permit terms and conditions, and monitoring actions described in the FEIS on pages 2-1 and 2-2 under "Actions Common to All Alternatives" will apply to all actions, as applicable. The Wilson Complex Fire Closure Decision dated May 25, 2006, which closes the Upper and Lower Fourmile Pastures until rehabilitation objectives are met, remains in effect until the BLM notifies the permittee that the closed area is re-opened to authorized livestock use. Other fire closure decisions from 2006 wildfires and beyond will carry the same effect on the grazing permit and livestock authorizations. ⁶

⁶ The amended BLM Grazing Regulations that became effective after the FEIS was issued (on August 12, 2006) recognize BLM's authority to respond as necessary to drought, fire and other resource conditions. (43 CFR 4110.3).

All other decision points not affected by the above modifications will remain as outlined in the April 19, 2002 FMUD.

8. 4 IMPLEMENTATION

Implementation of the modified multiple use decisions will occur over a number of years, beginning with issuance of the new 10-year grazing permits with the interim systems and the terms and conditions.

As the essential range improvements are constructed and monitoring indicates short-term objectives are being met, the expectation is the final grazing plans will be implemented.

Installation of range improvements will be dependent on the availability of funds. Priorities are developed through the long-term budgeting process: and in consultation with affected interests. Site-specific NEPA compliance for will be documented as locations are determined and following the establishment of the need for

the project. Surveys and consultations for specific resources (e.g., cultural resources, Native American concerns, nesting birds) will be conducted, where needed, prior to construction of ground-disturbing projects, as stated in the Management Framework (Appendix B) of the FEIS.

BLM will continue to collect actual use reports from the permittees, and to monitor and evaluate rangeland health conditions in accordance with priorities established by the Elko Field Office.

The three-final grazing decisions are subject to appeal in accordance the BLM Grazing Regulations at 43 OFR 4160.4 Pending resolution of any appeal(s), the changes to grazing management will become effective in May 2007 for the Sheep Allotment Complex and in March 2007 for the Big Springs and Owyhee allotments.